Western University Scholarship@Western

Electronic Thesis and Dissertation Repository

5-9-2023 2:30 PM

Regulatory Implications of Inadequately Designed Pimavanserin Drug Trials Published with Risk of Bias on Expedited Regulatory Approval Processes

Benson Law, Western University

Supervisor: Johnson, Andrew M., *The University of Western Ontario* Co-Supervisor: Smith, Maxwell J., *The University of Western Ontario* A thesis submitted in partial fulfillment of the requirements for the Master of Science degree in Health and Rehabilitation Sciences © Benson Law 2023

Follow this and additional works at: https://ir.lib.uwo.ca/etd

Part of the Neurology Commons, Patient Safety Commons, Psychiatric and Mental Health Commons, and the Quality Improvement Commons

Recommended Citation

Law, Benson, "Regulatory Implications of Inadequately Designed Pimavanserin Drug Trials Published with Risk of Bias on Expedited Regulatory Approval Processes" (2023). *Electronic Thesis and Dissertation Repository*. 9309. https://ir.lib.uwo.ca/etd/9309

This Dissertation/Thesis is brought to you for free and open access by Scholarship@Western. It has been accepted for inclusion in Electronic Thesis and Dissertation Repository by an authorized administrator of Scholarship@Western. For more information, please contact wlswadmin@uwo.ca.

Abstract

The objective of this retrospective critical appraisal study was to determine if the trials submitted to the Food and Drug Administration for the expedited approval of pimavanserin was of sufficient methodological quality to ascertain its safety and efficacy. After the general metrics of the trials were assessed, the Risk of Bias 2 tool and the PRagmatic Explanatory Continuum Indicator Summary tool were employed to evaluate the risk of bias and the design suitability of the trials. This study suggests that the decision to approve pimavanserin for the treatment of Parkinson's Disease Psychosis failed to meet the threshold of evidence normally required for FDA drug applications. It also revealed serious risks of bias with the pivotal trial that was foundational to pimavanserin's delineation as a breakthrough drug, and that the safety studies were also questionably designed. These findings highlight the need to continue monitoring pimavanserin and re-examine expedited drug approval processes.

Keywords

pimavanserin, Parkinson's Disease, Risk of Bias, Clinical Trials, FDA, Breakthrough Therapy.

Summary for Lay Audience

Clinical trials provide the foundational evidence used by the Food and Drug Administration (FDA) and other regulators to assess a drug's safety and efficacy before being approved for public use. Trials that are poorly designed may lead to a biased or overly favorable interpretation of a drug's safety and efficacy and may make it difficult for regulators to determine whether a new drug is safe to be used by patients. One of these new drugs, pimavanserin, was recently approved by the FDA for the treatment of Parkinson's Disease Psychosis. It was approved under a special program that allowed it to be approved more quickly than normal, and its application also received ongoing support from the FDA. These special programs are reserved for promising drugs that are meant to treat serious conditions. Shortly after, however, serious concerns were raised by clinicians and the broader academic community about the FDA's decision to approve pimavanserin. These concerns revolved around the higher rates of drug side effects, suggesting a potential risk to patients. The goal of this study was to conduct an assessment to determine if pimavanserin's trials were of sufficient quality, and to determine if there were any potential biases that may have compromised the FDA's initial assessment. Ultimately, this assessment revealed that the trials were of insufficient length and quality based on the FDA's own standards, and that the main clinical trial that formed the basis for approval was at risk of bias. In the context of clinical trials, a risk of bias means that the clinical data may present the drug optimistically, usually as more effective, or safer than it is. This is potentially dangerous because it means that clinicians may prescribe the drug without a true understanding of its effects, and this could harm patients. These issues with pimavanserin's clinical trials suggest that there could be a risk to patients, and that it

should continue to be monitored. This project also suggests that the FDA should consider changing how their special expedited programs assess and interpret the quality of clinical trials.

Acknowledgments

I would like to acknowledge the academic support from my supervisors, Dr. Maxwell Smith and Dr. Andrew Johnson, as well the support offered by Dr. Dianne Bryant. I want to also acknowledge the ongoing support of my family and friends.

Table of Contents	Table	e of	Con	tents
-------------------	-------	------	-----	-------

Ał	ostra	ct	ü
Su	mm	ary for	Lay Audienceiii
Ac	kno	wledgn	nentsv
Та	ble o	of Cont	entsvi
Li	st of	Tables	ix
Li	st of	Figure	S X
Li	st of	Abbrev	viations xi
Ch	apte	er 1	
1	Intr	oductio	n1
Ch	apte	er 2	
2	Bac	kgroun	d 6
	2.1	Burde	n of Disease Surrounding Parkinson's Disease and Psychosis
	2.2	Pimav	anserin A Novel Anti-Psychotic Medication10
		2.2.1	An Unmet Need within Existing Therapy Options
		2.2.2	Research and Development Timeline
		2.2.3	Expansion into Additional Indications 16
	2.3	FDA I	Regulation, Expedited Approvals, and Associated Ethical Concerns 18
		2.3.1	Regulatory Landscape, Investigational New Drug Applications, and New Drug Applications
		2.3.2	Expedited Approvals
		2.3.3	Summary of the Four Expedited Mechanisms
		2.3.4	Existing Issues and Past Recalls with Expedited Approvals
	2.4	Safety	Concerns with Pimavanserin
		2.4.1	Poor Clinical Evidence to Support Regulatory Approvals

		2.4.2	Increased Risk of Death, Adverse Events, and Serious Adverse Events	37
	2.5	Releva	ance to Canadians with Parkinson's Disease	38
C	hapte	er 3		41
3	Res	earch (Questions and Objectives	41
	3.1	Object	tive: The Importance of Clinical Trials Design	41
	3.2	Case H	Example: Biogen & Aduhelm (aducanumab-avwa)	42
С	hapte	er 4		44
4	Me	thodolo	gy	44
	4.1	Stage	One: Data Collection	44
	4.2	Stage	Two: Evaluation Using the RoB 2 and PRECIS-2 Tools	45
		4.2.1	Risk of Bias 2 Tool	46
		4.2.2	PRECIS-2 Tool	52
С	hapte	er 5		55
5	Res	ults		55
	5.1	Descri	ption of the Collected Studies	56
		5.1.1	Eligible Studies	56
		5.1.2	Description of the Eligible Studies	58
		5.1.3	Primary and Secondary Outcomes and Scales Utilized	61
	5.2	Risk o	f Bias 2 (RoB 2)	66
		5.2.1	Randomized Control Trials	68
		5.2.2	Safety Studies	73
		5.2.3	Single Arm Studies	75
	5.3	PREC	IS-2	77
		5.3.1	Randomized Control Trials	78
		5.3.2	Safety Studies	80

		5.3.3	Single Arm Studies	83			
Cl	napte	er 6		85			
6	Dis	cussion	l	85			
	6.1	Duty of	of Medical Experts: A Preface	86			
	6.2	Positiv	ve Trial Features	88			
	6.3	Prima	ry Concerns	89			
		6.3.1	Inadequate Number and Length of Trials	89			
		6.3.2	High Risk of Bias: Missing Data	91			
		6.3.3	Poor Trial Design and Poorly Understood Safety Profile	92			
	6.4	Contri	butions to the Literature	94			
6.5 Limitations							
	6.6	Next S	Steps	96			
	6.7	Conclu	usion	97			
Re	References						
С	Curriculum Vitae						

List of Tables

Table 1: Full list of Acadia Pharmaceuticals initiated or adjacent trials for pimavanserin
from the U.S. National Library of Medicine database14
Table 2: Summary of the four expedited approval mechanisms, their official definition,
origin of request, requirements, and benefits28
Table 3: Complete list of signaling questions from the RoB 2 tool
Table 4: Summary of study components. 60
Table 5: different endpoints used in the clinical trials, along with their frequency of use.
Table 6: Breakdown of missing data from each treatment arm. 71
Table 7: Breakdown of missing data from NCT0348288277

List of Figures

Figure 1: Incidence (per 100,000) of diagnosed parkinsonism, including PD, by sex and
age group, Canada, 2013-2014 form the Public Health Agency of Canada (Neurological
Health Charities Canada et al., 2014)
Figure 2: Timeline of pimavanserin's research, development, and regulatory approval17
Figure 3: Brief history of FDA regulatory and policy changes
Figure 4: Illustration the regulatory approval process (Food and Drug Administration,
2022c)
Figure 5: Hierarchy of benefits for the expedited review mechanisms
Figure 6: Algorithm for suggested judgement of risk of bias arising from the
randomization process (Sterne et al., 2019)47
Figure 7: Example of a completed PRECIS-2 tool (Aronson et al., 2008)53
Figure 8: An adapted PRISMA diagram of eligible studies
Figure 9: RoB 2 summary table66
Figure 10: Risk of bias as a percentage of total bias67
Figure 11: PRECIS-2 chart for RCTs78
Figure 12: PRECIS-2 chart for safety studies
Figure 13: PRECIS-2 chart for single-arm studies

List of Abbreviations

ADL: Activities of Daily Living ANDA: Abbreviated New Drug Application **BBW: Black Box Warnings** CGI: Clinical Global Impressions Scale CGI-I: Clinical Global Impressions - Improvement Scale CGI-S: Clinical Global Impressions - Severity Scale CMAJ: Canadian Medical Association Journal EQ-5D-5L: EuroQol-5-Dimensions-5 Levels FAERS: FDA Adverse Event Reporting System FDA: Food and Drug Administration FDAAA: Food and Drug Administration Amendments Act FDAMA: Food and Drug Administration Modernization Act of 1997 FDASIA: Food and Drug Administration Safety and Innovation Act GAO: US Government Accountability Office IND: Investigational New Drug **IRR:** Incidence Rate Ratio **ISMP: Institute for Safe Medication Practices** MDS: Movement Disorder Society MDS-UPDRS: Movement Disorder Society - Unified Parkinson's Disease Rating Scale mFSQ: Modified Functional Status Questionnaire NDA: New Drug Application PD: Parkinson's Disease PDAC: Psychopharmacologic Drugs Advisory Committee PDP: Parkinson's Disease Psychosis PDUFA: Prescription Drug User Fee Act PREICS-2: PRagmatic Explanatory Continuum Indicator Summary **PRO:** Patient Reported Outcomes **RCT:** Randomized control trial RoB 2: Risk of Bias 2 Tool SAPS: Scale for the Assessment of Positive Symptoms sDNA: Supplemental New Drug Application SCOPA-Sleep: Scales for Outcomes in Parkinson's Disease - Sleep SCOPA-DS: Scale for outcomes in Parkinson's Disease Sleep Scale - Daytime Sleepiness SCOPA-NS: Scale for outcomes in Parkinson's Disease Sleep Scale - Nighttime Sleep

UPDRS: Unified Parkinson's Disease Rating Scale

Chapter 1

1 Introduction

Parkinson's Disease (PD) is characterized by the progressive degeneration of dopaminergic neurons in the substantia nigra. The pathophysiological mechanisms of the disease have been extensively documented in the academic literature (Lang & Lozano, 1998). Common motor symptoms include tremors, bradykinesia, rigidity, balance, and posture impediments (Reich & Savitt, 2019). Additionally, there are non-motor symptoms such as cognitive decline (most notably related to memory and cognitive speed), speech problems (related to volume, pitch, and articulation), and psychosocial symptoms, including psychosis (Reich & Savitt, 2019). Parkinson's disease psychosis (PDP) poses a particular challenge as it can take a toll on both patients and care partners (Hermanowics & Edwards, 2015). PDP encompasses a wide variety of symptoms, including hallucinations that are usually visual in nature but that can involve other sensory domains, and delusions (Chen, 2017). While PDP may occur as a natural progression of PD, it is also a common side effect of levodopa, the primary drug intervention prescribed to manage PD (Moskovitz, Moses & Klawans, 1978), with approximately 50% of patients who take levodopa eventually developing PDP as a side effect (Hermanowics & Edwards, 2015). Although levodopa is not the sole cause of PDP, and it can manifest naturally over the disease course, research has indicated that levodopa substantially worsens PDP after one year of therapy (Friedman, 2010). Given its high prevalence, the management of PDP is important to PD patients, care partners, clinicians, and other stakeholders.

Traditional antipsychotics have been widely used to manage PDP, but their use has been linked to the progression of PD, rendering them counterproductive for symptom management (Chen, 2017). In 2016, pimavanserin tartrate (i.e., ACP-103), a selective inverse agonist of the 5-HT2A receptor marketed by Acadia Pharmaceuticals under the brand name Nuplazid, was approved by the United States Food and Drug Administration (FDA) for the management of PDP (Acadia Pharmaceuticals, 2016b).

The process by which industry-sponsored pharmaceuticals are authorized and made available to patients in the United States is a lengthy and complex process. This research and development process includes multiple pre-clinical and clinical trials with the goal of generating sufficient data to allow national regulatory authorities (e.g., FDA) to assess the safety and efficacy of the drug.

Sponsors (i.e., pharmaceutical companies) can pursue one of two submission pathways with the FDA in order to license their drug for use: standard or priority review, which relate to how long the FDA approval will take (Food and Drug Administration, 2018b). In addition to priority review, drugs may also be awarded Fast-Track, Breakthrough Therapy, and Accelerated Approval designations, which relate to the amount of support, communication, or trial design concessions that are offered by the FDA (Food and Drug Administration, 2018b). Pimavanserin's submission was approved with a Breakthrough Therapy designation within the Priority Review pathway because it is the first drug indicated specifically for the treatment of PDP (Acadia Pharmaceuticals, 2014; Acadia Pharmaceuticals, 2015).

Drug safety is less of a binary question (i.e., determining whether a drug is safe or unsafe), and more of a relative exercise in which the drug's potential therapeutic benefit is weighed with an established safety and adverse events profile. With a standard review, drugs usually go through a lengthy and extensive review by the FDA, after which they receive conditional approval for market authorization, or alternatively, are rejected. However, with the alternative and expedited regulatory review streams introduced in 1992, 1997, and 2012 by the FDA, sponsors can receive early market authorization with limited data. This means that if there are adequate justifications, sponsors can receive market authorization without comprehensive stage three clinical trials or with the use of less rigorous surrogate outcomes, amongst other concessions (Kepplinger, 2015). Such justifications are usually grounded in the view that a drug will address an unmet need, treat serious conditions, or demonstrate promising benefit over existing lines of therapy (Food and Drug Administration, 2018b; Kepplinger, 2015). There are well warranted concerns with these deviations from traditional approval mechanisms, as this means that drugs may be approved without the regularly applied checks and balances, contrary to the FDA's public messaging (Frakt, 2018; Redberg, 2015; Kim & Prasad, 2015). Determining which drugs are approved in this way depends upon interpretations of these justifications by sponsors and by the FDA.

Shortly after pimavanserin received expedited market authorization by the FDA through the Breakthrough Therapy designation and Priority Review pathway, the Institute for Safe Medicines Practices (ISMP) published concerning findings based on adverse events posted in the FDA Adverse Event Reporting System (FAERS) (Institute for Safe Medication Practices, 2017). The FAERS system aggregates adverse drug events reported by consumers and clinicians after a drug has been approved for use (Food and Drug Administration, 2021b). Based on their preliminary investigation, the ISMP uncovered that there was a limited body of evidence to support pimavanserin's approval. They also cited emerging evidence pointing to concerning side effects not previously reported, including hallucinations, confusion, and death (Institute for Safe Medication Practices, 2017).

As of April 2021, the FDA has rejected the sponsor's (i.e., Acadia Pharmaceuticals) request to expand pimavanserin for use in dementia-related psychosis (Acadia Pharmaceuticals, 2021). Although the sponsor has stated that they stand behind Nuplazid's "robustly positive results", there is growing concern regarding pimavanserin's safety and efficacy (Acadia Pharmaceuticals, 2022a; Acadia Pharmaceuticals, 2022b Hwang et al., 2021).

This study presents a randomized control trial (RCT) critical appraisal aimed at examining: 1) the quality of the clinical trials design based on the degree of bias as determined by the RoB 2 tool; and 2) the overall body of clinical trial evidence supporting pimavanserin's approval. Doing so can help to 1) better interpret the results of pimavanserin's drug trials; 2) inform American and other regulatory decisions; and 3) improve future expedited approval practices. As it is conceptually unfeasible to provide an absolutely "objective" interpretation of trial design and data, the results of this study will be supported with validated assessment tools and will be considered in relation to the FDA's own regulatory decision. In this way, this study aims to provide a better understanding of whether there should have been other considerations, such as the potential risk of bias in the clinical trials, when approving this potentially contentious pharmacotherapy.

The structure of this paper will first proceed with an introduction of the burden of disease for PDP and pimavanserin's research and development timeline. A description of the FDA's regulatory approval process, including the four different expedited pathways and their associated issues, will then be provided to give readers a foundational understanding of how the FDA operates. Following this introduction of the FDA, the issues associated with pimavanserin, and its approval will then be discussed. All this information will cumulate into Chapter 3, which will introduce the research question and describe the rationale of the study by emphasizing the importance of clinical trials design and rigor. The methods, and the justification for their employment, will then be described in Chapter 4. The results and their interpretation will be described in Chapter 5 and 6, respectively.

Chapter 2

2 Background

The purpose of this chapter is to provide readers with an understanding of the prevalence of PD and PDP, as well as illustrate the unmet need that pimavanserin was advertised to meet. This chapter will also provide readers with an understanding of pimavanserin's research and development timeline leading up to their regulatory approval. This will be followed by a review of the FDA's regulatory approval process, their expedited pathways, and the associated issues. Finally, pimavanserin's safety concerns which underly the need for this study, will be discussed.

2.1 Burden of Disease Surrounding Parkinson's Disease and Psychosis

PD is a debilitating neurodegenerative disease, primarily characterized by the death of dopaminergic neurons in the substantia nigra, leading to clinical symptoms including tremors, rigidity, and akinesia (Lang & Lozano, 1998). Discussions surrounding the specific pathophysiological mechanisms of PD are beyond the scope of this project, and these mechanisms have already been well documented in the literature by landmark articles like Lang & Lozano (1998). Of importance, however, is the increasing financial and healthcare burden that PD poses to patients, care partners, and clinicians. This is

important because the growing burden, if left unaddressed, may mean that the patients do not receive the appropriate treatment options.

The incidence and prevalence of PD is on the rise. Based on a 2016 systematic review and meta-analysis, the global incidence and prevalence of PD is increasing for both males and females (Hirsch et al., 2016). A separate 2016 systematic review of epidemiological data confirmed that from 1990 to 2016, Canada reported a 43.0% (95% CI: 16.5% to 67.0%) increase in the prevalence of PD and 44.5% (95% CI: 15.1% to 69%) increase in the burden of PD, measured in disability-adjusted life-years (Dorsey et al., 2018).

This increase is primarily caused by the growing aging population of Canada. Based on the Population Health Model, a series of microsimulations performed by Statistics Canada and the Public Health Agency of Canada, Canada's demographic distribution of adults 65+ will shift from 15% in 2011 to approximately 23% in 2031, substantially increasing the incidence rate of PD (Neurological Health Charities Canada et al., 2014). Combined with data from the Public Health Agency of Canada shown in *Figure 1* there is, and will continue to be, a growing impact of PD on our Canadian healthcare system as our population ages.



Figure 1: Incidence (per 100,000) of diagnosed parkinsonism, including PD, by sex and age group, Canada, 2013-2014 form the Public Health Agency of Canada (Neurological Health Charities Canada et al., 2014).

The burden of disease for patients, their care partners, and on the healthcare system cannot be understated. The management of PD is arduous, and involves substantial financial, emotional, and physical resources.

The most recent aggregated data from a 2007 Canadian Institute for Health Information report detailed that between 2000 and 2001, the total direct and indirect costs related to PD totaled \$201.9 million and \$244.9 million, respectively (Canadian Institute for Health Information, 2007). When compared to other neurological conditions, PD is the third most economically intensive neurological condition, ranking only behind epilepsy and dementias in healthcare spending (Canadian Institute for Health Information, 2007). As well, researchers from the British Columbia Administrative Data Project discovered that patients with PD have the highest prescription use (i.e., highest per capita dispensed prescription days), when compared to other neurological conditions such as Alzheimer's Disease and Huntington's Disease, amongst others (Neurological Health Charities Canada et al., 2014). The costly prescription drug regimen used to address the symptoms of PD presents an ongoing challenge because multiple concurrent lines of therapy are often implemented to help manage both motor and non-motor symptoms.

One of the most taxing non-motor symptoms for patients and care partners is PDP, which occurs in 50% of patients over the course of their disease experience (Hermanowics & Edwards, 2015). PDP is a symptom that presents secondary to treatment with levodopa, the first-line treatment for PD, and is often used as an indicator for a diagnosis of PD (Grimes et al., 2019). However, it should also be noted that PDP can on rare occasion present without treatment with levodopa, but research has indicated that levodopa substantially worsens PDP after one year of therapy (Friedman, 2010). PDP may also cause secondary incidents such as falls, infections, and other concerns that put the patient at increased risk for hospitalization, in-home care, or permanent placement within a long-term care home (Kalilani et al., 2016).

Given the prolonged demands of care associated with PDP, there is a negative impact on care partner stress, depression, and quality of life (Schrag et al., 2006). As a result, the management of PD and PDP is sometimes concurrently supplemented with traditional antipsychotics, which are defined as drugs that primarily act on the dopaminergic system. Unfortunately, the use of these first-generation antipsychotics (i.e., dopamine receptor antagonists) may lead to the worsening of PD motor symptoms, making it counterproductive to the overall management of PD symptoms (Chen, 2017). As such, there is a strong clinical need for atypical antipsychotics (i.e., higher affinity for serotoninergic receptors than dopaminergic receptors) that can minimize the presentation of PDP without hastening the progression of PD.

2.2 Pimavanserin A Novel Anti-Psychotic Medication

2.2.1 An Unmet Need within Existing Therapy Options

In 2016, pimavanserin tartrate (i.e., ACP-103), a selective serotonin inverse agonist of the 5-HT2A receptor marketed under the brand name Nuplazid, was approved by the FDA (Acadia Pharmaceuticals, 2016b). There are other treatment options that are currently employed by clinicians, but they are not specifically approved for use on patients with PDP (i.e., used off-label). At the time of this project, pimavanserin remains the only medication specifically approved for the treatment of PDP in the United States.

The management of PDP continues to be severely challenging. As demonstrated in a 2021 study of PD patients in the United States, the rates of initial and overall discontinuation for antipsychotics were at 38.6% and 61.4%, respectively (Pham Nguyen et al., 2021). These figures were more pronounced in patients using traditional antipsychotics with dopamine receptor blocking activity, including quetiapine, aripiprazole, risperidone, and olanzapine (Pham Nguyen et al., 2021). This discontinuation can be attributed to the worsening of parkinsonism caused by interactions with dopaminergic receptors, supporting the notion that there is a clinically unmet need for this indication (Pham Nguyen et al., 2021).

Prior to pimavanserin, clozapine was also suggested as a viable alternative for the management of PDP (Thomas & Friedman, 2010). As a non-dopaminergic antipsychotic,

clozapine demonstrated promising short-term and long-term efficacy in a limited number of published samples (Parkinson Study Group, 1999; Pollak et al., 2004). However, it is rarely used because of logistical barriers, most notably the stringent monitoring requirements that are in place because of the increased risk this medication poses for agranulocytosis, cardiovascular and respiratory events, and other adverse events (Fernandez et al., 2004).

As the first medication in the United States approved for the management of PDP advertised to be without many of the drawbacks from other first-generation antipsychotics (i.e., having dopaminergic effects), pimavanserin received interest from clinicians as it was seen as potentially serving an unmet need within existing lines of therapy (Grimes et al., 2019).

2.2.2 Research and Development Timeline

Because mandatory reporting of clinical trials was not in effect prior to September 27, 2007 (National Library of Medicine, 2020), the most complete sourcing of pimavanserin's development timeline before 2007 is contained within Acadia Pharmaceuticals' patent filing to the World Intellectual Property Organization, filed in November 2008 (Vanover et al., 2008). Within the patent filing, there is mention of two studies, one Phase 1 and one Phase 2 study to investigate the pharmacokinetic properties of pimavanserin, along with dose-response studies with fasted and fed healthy human subjects without PDP (Vanover et al., 2008).

In addition to these two studies, a variety of Phase 2 and 3 studies were conducted between March 2004 to March 2008 before the patent was filed in November 2008. These were made public in the clinicaltrials.gov database. These studies include a Phase 2 open label safety study, a Phase 2 interventional study, a Phase 3 safety and tolerability study, and two additional Phase 3 safety and efficacy studies. An aggregated list of Acadia Pharmaceutical initiated or adjacent trials for pimavanserin, spanning 2004 to 2024, has been included in *Table 1*.

Study Identifier (NCT /ACP)	Study Title	Phase	N	Start	Completed	Results Posted	Completed
NCT01518309 ACP-103-010	An Open-label Safety Study of Pimavanserin in Parkinson's Disease Patients	Phase 2	39	November 17, 2004	May 2, 2013	November 23, 2020	Yes
NCT00087542 ACP-103-006	Treatment of Hallucinosis/ Psychosis in Parkinson's Disease by an Investigational Drug	Phase 2	60	March 2004	December 2005	N/A	Yes
NCT03482882 ACP-103-048	Safety and Efficacy of Pimavanserin in Adults With Parkinson's Disease and Depression	Phase 2	47	March 9 2018	July 9, 2019	August 31, 2020	Yes
NCT00550238 ACP-103-015	A Study of the Safety and Tolerability of Pimavanserin (ACP-103) in Patients With Parkinson's Disease Psychosis	Phase 3	459	July 2007	May 30, 2018	June 24, 2019	Yes
NCT01174004 ACP-103-020	Study of the Safety and Efficacy of Pimavanserin in Patients With Parkinson's Disease Psychosis	Phase 3	199	July 2010	November 2012	March 26, 2014	Yes
NCT00477672 ACP-103-012	A Study of the Safety and Efficacy of Pimavanserin (ACP-103) in Patients With Parkinson's Disease Psychosis	Phase 3	298	June 2007	June 2009	March 26, 2014	Yes

NCT00658567 ACP-103-014	A Study of Safety and Efficacy of Pimavanserin (ACP-103) in Patients With Parkinson's Disease Psychosis	Phase 3	123	March 2008	December 2009	September 9, 2014	Yes
NCT04292223 ACP-103-063	Open-Label Study With Pimavanserin on Activities of Daily Living in Subjects With Parkinson's Disease Psychosis	Phase 4	29	February 10, 2020	April 26, 2022	N/A	Yes
Terminated / No Updates							
NCT02762591 ACP-103-036	Expanded Access of Pimavanserin for Patients With PD Psychosis					N/A	Approved for Marketing
NCT03152292 ACP-NIS-001	The INSYTE (Management of Parkinson's Disease Psychosis in Actual Practice) Study		764	March 30, 2017	March 8, 2021	N/A	Terminated

Table 1: Full list of Acadia Pharmaceuticals initiated or adjacent trials for

pimavanserin from the U.S. National Library of Medicine database.

In September 2014, the sponsor Acadia Pharmaceuticals received a Breakthrough Therapy designation for pimavanserin, allowing for special expedited review and preferential access to FDA supports (Acadia Pharmaceuticals, 2014). The following year, in September 2015, Acadia Pharmaceuticals submitted a New Drug Application (NDA) to the FDA, in which they sought regulatory approval under the Priority Review pathway (Acadia Pharmaceuticals, 2015). In November 2015, the FDA approved pimavanserin for the priority review pathway, shortening the expected review time by 40%, from 10 months to 6 months (Acadia Pharmaceuticals, 2015).

During the FDA's internal review process, Paul Andreason, a member of the FDA's Division of Psychiatry Products and the designated medical reviewer for the pimavanserin submission, completed his extensive 173-page review of pimavanserin, where he explicitly provided a strong "do-not-approve" recommendation, citing "unacceptably increased, drug- related, safety risk of mortality and serious morbidity" (Andreason, 2016). Despite his review, which was released internally in February 2016, the FDA's Psychopharmacologic Drugs Advisory Committee (PDAC) voted 12 to 2 in support of pimavanserin's efficacy and safety profile, leading to pimavanserin's regulatory approval in April 2016 (Acadia Pharmaceuticals, 2016a). Pimavanserin was approved based primarily on efficacy data from one clinical study (NCT01174004, ACP-103-020) that had a sample size of 199, along with sparse safety data from previous and concurrent trials (Food and Drug Administration, 2016). The decision to approve pimavanserin caused internal division amongst key reviewers within the FDA (Mathis et al., 2017), and sparked a broader public investigation and conversation about pimavanserin's safety (Ellis & Hicken, 2018).

2.2.3 Expansion into Additional Indications

Acadia Pharmaceuticals is currently (as of 17 April 2023) advocating for expanded regulatory approval of pimavanserin for dementia-related psychosis and Alzheimer's disease psychosis (Acadia Pharmaceuticals, 2022c; Soogrim et al., 2021). In September 2019, Acadia Pharmaceuticals stopped their pivotal HARMONY trial for the management of dementia-related psychosis, citing that they had achieved the primary endpoint for the treatment of dementia-related psychosis (Tariot et al., 2021). In June 2022, the FDA's PDAC committee voted 9 to 3 against pimavanesrin's efficacy and safety profile for the treatment of Alzheimer's disease psychosis (Acadia Pharmaceuticals, 2022a). In August 2022, the FDA rejected Acadia Pharmaceutical's Supplemental New Drug Application (sNDA) for the treatment of Alzheimer's disease psychosis (Acadia Pharmaceuticals, 2022b). Although this is not specifically related to PDP, these evolving updates may provide contextual insight into FDA's changing attitude towards pimavanserin. Without additional insight into the FDA's specific reasoning, it is difficult to pinpoint the exact reason behind this changing attitude towards pimavanserin as a Breakthrough Therapy product. It may be caused by a combination of factors, including response to negative press, a more careful examination of the clinical data, or other variables that have not yet been made public.

Figure 2 presents a timeline that illustrates pimavanserin's development and eventual market approval by the FDA. The data was amalgamated from a variety of sources, including US Library of Medicine clinical trials database and patent filings (U.S. National Library of Medicine, 2005; U.S. National Library of Medicine, 2014; U.S. National Library of Medicine, 2017a; U.S. National Library of Medicine, 2017b; U.S. National Library of Medicine, 2019; U.S. National Library of Medicine, 2020a; U.S.

National Library of Medicine, 2022; Vanover et al., 2008).



Figure 2: Timeline of pimavanserin's research, development, and regulatory approval.

2.3 FDA Regulation, Expedited Approvals, and Associated Ethical Concerns

2.3.1 Regulatory Landscape, Investigational New Drug Applications, and New Drug Applications

The drug approval process is lengthy, born out of numerous safety incidents that prompted the US government to act on threats to consumer and patient safety. Most notably, the Elixir Sulfanilamide incident of 1937 and the Thalidomide incident of the 1950s and 1960s, amongst other tragedies, prompted regulatory and policy changes (Food and Drug Administration, 2019). A timeline of these developments has been included as *Figure 3*.



Figure 3: Brief history of FDA regulatory and policy changes.

For a drug to obtain regulatory approval in the United States, investigators must first conduct in-vitro or computational studies for potential new compounds. When a promising compound is identified, sponsors submit an Investigational New Drug (IND) application to the FDA to seek approval for shipping novel drug compounds across state lines for clinical trials (Food and Drug Administration, 2022d). Once sponsors believe they have sufficient evidence from Phase 1-3 clinical trials to undergo the FDA's review process, they submit an NDA (Food and Drug Administration, 2022b). Upon approval, drugs can be marketed in the United States.

Several NDA variations exist based on the type of drug and their stage in marketing. For drugs that are already approved, sponsors may submit a sNDA to expand the number of indications, labelling, ingredients, or packaging, as with pimavanserin's attempted expansion into dementia-related psychosis (Food and Drug Administration, 2017). With generic drugs, sponsors submit an abbreviated NDA (ANDA) to the FDA without the need for animal or human clinical studies (Food and Drug Administration, 2022a). For biologic products, sponsors are required to submit a Biologic License Application as per the regulations stated in the Public Health Service Act (Food and Drug Administration, 2020).

Once drug approval applications are submitted, the application undergoes review by the FDA's review team. This process includes individual assessments by each member of the review committee, who possess technical and clinical expertise (Food and Drug Administration, 2018a). This initial review will also include routine visits to the clinical sites by FDA inspectors to look for "evidence of fabrication, manipulation, or withholding of data" (Food and Drug Administration, 2018a). Additional review support can be provided by expert review committees, as with the PDAC in pimavanserin's regulatory submission (Food and Drug Administration, 2018a). Once each member of the FDA's review committee has completed their assessment, an aggregated recommendation is issued by the division director based on a risk-benefit assessment. However, there are no specific documents outlining the exact process that guide this risk-benefit assessment and may depend on the individual expertise of the director in charge of that disease space.

Critics have also cited that these decision-making processes are not transparent enough, especially when it comes to drugs that are denied for market authorization (Sharfstein et al., 2017). In cases where drugs are denied for regulatory approval, information is often withheld from the public because much of the clinical data is considered proprietary (Sharfstein et al., 2017). As such, the clinical data and specific reasoning within the appraisal process is often kept private between the FDA and the sponsor. These criticism and proposed recommendations for increased transparency within FDA processes has been extensively explored by Sharfstein et al. (2017) and should be referenced. *Figure 4* has been included from the FDA's Center for Drug Evaluation and Research to illustrate an example of a drug's progression through the different regulatory approval stages.



Figure 4: Illustration the regulatory approval process (Food and Drug Administration, 2022c).

2.3.2 Expedited Approvals

For drugs that treat a serious condition, satisfy an unmet need, or demonstrate superiority over existing treatment options, the FDA created four expedited pathways as part of ongoing policy changes to bring promising therapies more quickly to patients. These four are the Priority Review pathway, the Breakthrough Therapy designation, the Fast Track designation, and the Accelerated Approval designation (Food and Drug Administration, 2018b). These changes were originally born out of the need to expedite the regulatory process to better manage the novel AIDS/HIV pandemic of the 1980s (Kepplinger, 2015).

2.3.2.1 Priority Review Designation

Prior to the 1992 changes, drugs were prioritized for review under the A, B, and C potential therapeutic effect classifications, which denoted important, modest, or limited therapeutic benefit, respectively (Kepplinger, 2015). This classification system would be revised as part of the Prescription Drug User Fee Act (PDUFA), which re-classified the A and B classifications into Priority Review (P), and C designation into Standard Review (Kepplinger, 2015). The original top priority (AA), and subpar drug (E) were retained from earlier policy changes (Kepplinger, 2015). The PDUFA allowed the FDA to levy new fees as part of sponsor submissions to hire new staff to meet the shortened review periods (Kepplinger, 2015).

With these new changes, standard reviews were now conducted in 10 months and priority reviews conducted in only 6 months (Food and Drug Administration, 2018a). Priority reviews are granted if, in the view of the FDA, the candidate drug provides superior treatment, diagnostic, or prevention capabilities (Food and Drug Administration, 2018a). This can be demonstrated by evidence of increased efficacy, demonstration of reduction in adverse reactions, improved compliance rates amongst patients, or improved safety and efficacy within new subpopulations (Food and Drug Administration, 2018a).

2.3.2.2 Accelerated Approval Designation

As part of ongoing changes during this time, the Code of Federal Regulations pertaining to the FDA were also amended (Kepplinger, 2015). These new changes allowed drugs that were indicated for the treatment of serious conditions with an existing unmet need to submit an NDA using unvalidated or novel surrogate endpoints, yielding immature data in lieu of more mature data (Food and Drug Administration, 2018a). The definition of 'unmet need' is only formally defined by the FDA within the Fast-Track designation and not the Accelerated Approval designation. The formal definition of unmet need is introduced under 2.3.2.3.

2.3.2.3 Fast-Track Designation

In 1997, as part of the Food and Drug Administration Modernization Act of 1997 (FDAMA), the Fast-Track designation was created (Kepplinger, 2015). Drugs that demonstrate an unmet need and are meant to treat serious conditions, especially those conditions that would deteriorate if left untreated, would be eligible for this designation (Food and Drug Administration, 2018a). Serious conditions are a matter of judgement and not formally defined by the FDA (Food and Drug Administration, 2018a). Unmet need is defined as a drug that shows superior effectiveness, avoids serious side effects, improves diagnosis leading to improved outcomes, decreases toxicity that would otherwise lead to treatment discontinuation, or addresses an emerging or anticipated public health need (Food and Drug Administration, 2018a).

As part of this Fast-Track designation, sponsors are eligible for more frequent meetings with FDA regulators who would aid with their submissions, and also makes them eligible for the Priority Review pathway and Accelerated Approval designation, as well as a rolling review, which enables partial NDA submissions in lieu of a complete NDA submission (Food and Drug Administration, 2018a). One of the most notable drugs to receive the fast-track designation was Vioxx, a selective nonsteroidal antiinflammatory drug that would go on to be recalled, but not before it caused substantial harm to patients, in the form of 88,000-140,000 additional cases of serious coronary heart disease in the United States (CMAJ, 2005; Graham et al., 2005). Subpoenaed documents later revealed that the sponsor (Merck) was aware of these risks during drug development and marketing (Whitstock, 2017).

2.3.2.4 Breakthrough Designation

In 2012, through the Food and Drug Administration Safety and Innovation Act (FDASIA) passed by Congress, the Breakthrough Therapy designation was created in part from pressure exerted by the Brookings Institute and Friends of Cancer Research; think tanks that are both partially funded by pharmaceutical companies (Brookings Institute, 2021; Friends of Cancer Research, 2020). The designation, as defined by the FDA,

"... is intended to facilitate and expedite the development of those drugs that receive designation and involves a resource commitment from FDA to provide early and frequent advice, conduct multidisciplinary meetings involving senior managers, and when appropriate, expedite the review of resultant marketing applications" (U.S. Department of Health and Human Service et al., 2022).

Once granted approval through this new designation, drugs are conferred all the benefits of the Fast-Track designation, in addition to intensive guidance from the FDA, and designated support from senior FDA managers (Food and Drug Administration, 2018a). After receiving this designation, sponsors try to leverage these benefits and seek regulatory approval. Pimavanserin indicated for the treatment of both PDP and dementia-
related psychosis was supported through this designation, concurrent with the Priority Review pathway.

2.3.3 Summary of the Four Expedited Mechanisms

Although the requirements to engage the four mechanisms are similar in that they all need to satisfy an unmet need or a serious medical condition, they are four distinct mechanisms with tiered benefits, with the Breakthrough Therapy designation conferring the most support from the FDA, as illustrated in *Figure 5*. A table summarizing the key requirements and features has also been included below as *Table 2*. An in-depth analysis of these pathways is beyond the scope of this project, but the work of Kepplinger (2015) and Cox et al. (2020) can be referenced for further details.



Figure 5: Hierarchy of benefits for the expedited review mechanisms.

Mechanism (Year Introduced)	FDA Definition	Origin of Request /Stage of Request	Requirements	Features
Priority Review Pathway (1992)	"A Priority Review designation will direct overall attention and resources to the evaluation of applications for drugs that, if approved, would be significant improvements in the safety or effectiveness of the treatment, diagnosis, or prevention of serious conditions when compared to standard applications" (Food and Drug Administration, 2018a).	-Determined by the FDA after receiving NDA from sponsor -Post-NDA	-Evidence under review should <i>demonstrate</i> <i>significant safety and</i> <i>efficacy</i> compared to existing treatment, diagnosis, or prevention options	- Reduces the review period by 40%, from 10 to 6 months
Accelerated Approval Designation (1992)	"in 1992 FDA instituted the Accelerated Approval regulations. These regulations allowed drugs for serious conditions that filled an unmet medical need to be approved based on a surrogate endpoint. Using a surrogate endpoint enabled the FDA to approve these drugs faster" (Food and Drug Administration, 2018a).	-Discussed in advance between the FDA and the sponsor -Prior to clinical studies	-Drug must demonstrate an effect on the surrogate or intermediate outcome -Sponsor must demonstrate that the surrogate or intermediate outcome can predict the primary clinical outcome -Post-market confirmatory Stage 4 trials must be conducted	-Allows for the use of a surrogate or intermediate endpoint instead of a primary clinical endpoint

Fast-Track Designation (1997)	"Fast track is a process designed to facilitate the development, and expedite the review of drugs to treat serious conditions and fill an unmet medical need. The purpose is to get important new drugs to the patient earlier. Fast Track addresses a broad range of serious conditions" (Food and Drug Administration, 2018a).	-Requested by the sponsor -Pre-IND and anytime thereafter	 -Early-stage application: Demonstrate benefit using nonclinical mechanism or early pharmacological data -Late-stage application: Clinical data supporting <i>potential for</i> <i>improvement</i> over existing lines of therapy 	-More frequent meetings with the FDA -More frequent communications with the FDA
Breakthrough Therapy Designation (2012)	"Breakthrough Therapy designation is a process designed to expedite the development and review of drugs that are intended to treat a serious condition and preliminary clinical evidence indicates that the drug may demonstrate substantial improvement over available therapy on a clinically significant endpoint(s)" (Food and Drug Administration, 2018a).	-Requested by the sponsor -Pre-IND and anytime thereafter	-Early clinical evidence demonstrates <i>substantial</i> improvement on a <i>clinical endpoint</i>	-All "fast-track" features -Support from senior FDA managers -Intensive drug development guidance early in the development process

Table 2: Summary of the four expedited approval mechanisms, their officialdefinition, origin of request, requirements, and benefits.

2.3.4 Existing Issues and Past Recalls with Expedited Approvals

These mechanisms were created with the express purpose of hastening the drug approval process. Despite the FDA's prescribed messaging that expedited drugs undergo the same rigorous review process as standard reviews, the evidence from the literature presented in the following sections challenges this view.

It should be noted that the purpose of highlighting the shortfalls of expedited approvals is not an attack on regulatory processes as a whole. It would be unfeasible to develop a regulatory system where 100% of drugs are completely safe. So, although regulatory processes are not without their faults, the issues being illustrated here are specifically with expedited approvals, and the regulatory concessions granted by the FDA, that pose an additional level of risk that may not be warranted for patients. While patients and clinicians might accept a greater risk to bring a novel therapy more quickly to market, reasonable disagreement exists as to how much risk should be tolerated.

As mentioned previously, Vioxx is a landmark example of a drug that was approved using these pathways, which has caused substantial harm to patients, their families, communities, and for healthcare systems. Vioxx, amongst other examples, were approved with inadequate clinical trial data and inaccurate marketing strategies, leading to inadequately studied products being tested on the public (CMAJ, 2005).

Several issues have been uncovered while conducting the background research for this project. These issues within the literature concerning these four expedited approval mechanisms include: 1) the lack of confirmatory trials leading to high rates of market withdrawal (Mahase, 2021); 2) the increased risk of black box warnings (Liang, 2002); and 3) the poor reliance on surrogate outcomes (Lenzer & Brownlee, 2021).

2.3.4.1 Lack of Confirmatory Trials and Untimely Market Withdrawal

When drugs receive expedited approval, sponsors are expected to conduct Phase 4 confirmatory post-market studies to study the real-world consequences for patients. A review conducted by Mahase (2021), however, found that of the 253 drugs approved by the accelerated streams over a 28-year period since 1992, 112 (44.27%) expedited drugs were not confirmed as clinically effective using Phase 4 studies. Mahase (2021) further notes that of those 112 drugs, many remained on the market despite poor effectiveness, with only 16 having been withdrawn, and one ineffective drug remaining on the market for as long as 12 years. More than two thirds of the drugs that had been on the market for over five years had not conducted confirmatory trials. As well, many of the sponsors failed to provide a response upon investigation by the authors of that review as to why confirmatory trials had not been conducted (Mahase, 2021).

Additional research with expedited oncology drugs has discovered that long-term studies after receiving market authorization are only performed in two thirds of cases and are usually delayed by an average of four years after market authorization (Johnson et al., 2011). In essence, this means that sponsors are not upholding their duty to conduct confirmatory trials in a timely manner; ineffective and possibly dangerous drugs are remaining on the market for years, putting patients at risk. Both Mahase (2021) and Johnson et al. (2011) echoed an earlier 2015 study conducted by the US Government Accountability Office (GAO), which found that sponsors were submitting and receiving approval from the FDA for a large number of applications under the Fast-Track and Breakthrough Therapy designations (US Government Accountability Office, 2015). For the Fast-Track designation applications submitted between 2006 and 2014, 525 of 772 (68.00%) requests were approved. For the Breakthrough Therapy designation applications submitted between 2012 and 2014, 71 of 225 (31.56%) requests were approved (US Government Accountability Office, 2015). Despite a large volume of expedited approvals, the GAO report concluded that "[t]he FDA lacks reliable, readily accessible data on tracked safety issues and post-market studies needed to meet certain post-market safety reporting responsibilities and to conduct systematic oversight" (US Government Accountability Office, 2015). Combined with the apparent unwillingness of sponsors to fulfill their post-market obligations, this lack of systemic oversight could put patients at risk.

While the FDA and sponsors share responsibility for these issues, it is important to recognize that the lack of adequate post-market surveillance is particularly dangerous because it shifts the onus of drug safety from a pre-regulatory approval setting (i.e., clinical trials) to a post-market surveillance setting (i.e., on the market).

Ultimately, Mahase (2021), Johnson et al. (2011), and the report from the U.S Government Accountability Office illustrate the first of many problems, which is that the FDA has fostered a regulatory environment that approves an increasing volume of drugs under expedited pathways without ensuring proper post-market surveillance mechanisms are in place to monitor safety. The remaining problems will be discussed in section 2.3.4.2 and section 2.3.4.3.

2.3.4.2 Increased Utilization of Black Box Warnings, Withdrawals, and Safety Communications

Black box warnings (BBW) are the most serious warning that can be prescribed by the FDA (Liang, 2002). These warnings, if assigned by the FDA, mandate sponsors to provide restrictions and warning of serious injury or death on all advertising, labelling, and promotional products associated with the drug product (Liang, 2002). Drugs that are part of accelerated approvals and indicated for psychiatric conditions were more likely to have BBWs, risk of safety events, product withdrawals, and safety communications (Downing et al., 2017). The increased risk of post-market safety events is quantifiable with accelerated approvals having an Incidence Rate Ratio (IRR) = 2.20 (95% CI, 1.15-4.21; P = .02), and psychiatric indications having an IRR = 3.78; (95% CI, 1.77-8.06; P < .001) (Downing et al., 2017).

Although these labels should be meant to alert prescribers of the potential risks, research has suggested that compliance with boxed warnings have mixed effects on prescribing patterns (Wagner et al., 2006), and in certain patient cases, no impact at all (Soumerai et al., 1987).

In a 2006 retrospective study of 929,958 health plan members in the United States, it was discovered that 41.7% of patients received a drug with a BBW that applied to their specific population (Wagner et al., 2006). Half of the compliance infractions were caused by inadequate monitoring, meaning that even though monitoring requirements are required by the BBW, clinicians are often non-compliant because they do not conduct or follow-up with the necessary monitoring tests (Wagner et al., 2006). Poor compliance was also found to be more pronounced amongst older adult populations (Wagner et al., 2006). Although there could be many valid reasons to explain the high number of BBW prescriptions and poor compliance, Wagner et al. (2006) ultimately suggests that the efficacy and potential impact of BBW on patients is debatable.

A study examining drug approvals between 2001 and 2010 found that drugs approved through the accelerated approval pathways and indicated for psychiatric conditions were at increased risk of post-market safety events, including withdrawals and the release of safety information (Downing et al., 2017). However, the authors did not find risk associated with the priority review stream (Downing et al., 2017). It should also be noted that the Breakthrough Therapy designation and Fast-Track designation were not assessed in this study; only the Priority Review pathway and Accelerated Approval designation were examined (Downing et al., 2017).

2.3.4.3 Poor Implementation Surrounding Surrogate Outcomes

The use of surrogate outcomes is approved under the Accelerated Approval designation, as described in Table 2. Surrogate outcomes, such as reduced tumor size, are substitute metrics that reduce the cost and length of clinical trials by reliably predicting clinical outcomes such as death (Fleming & DeMets, 1996). Kemp and Prasad (2017) however, has cited issues regarding its applications, specifically that surrogate outcomes are used without adequate statistical correlation to clinical endpoints, psychometric

validation, and that they are overused by sponsors to reduce the research and development timeline, yielding cost savings. Again, as stated previously, the use of surrogate endpoints is predicated on the expectation that confirmatory trials will be conducted after market launch to validate the surrogate endpoints, which as stated previously, do not always happen (Mahase, 2021).

A recent paper by Lenzer and Brownlee (2021) criticized the FDA for defending the Accelerated Approval pathway by citing rare success cases using surrogate outcomes but ignoring the unreliable nature and potential impact from an efficacy and health economic perspective. Lenzer and Brownlee (2021) identified four main issues with surrogate markers: 1) noncausal associations; 2) multiple casual pathways; 3) insensitivity; and 4) unintended outcomes.

Of the drugs that were approved through the Accelerated Approval pathway between 1992 to 2017, only 39 studies conducted completed confirmatory trials, of which 19 used the same surrogate outcomes as their initial trials (Gyawali et al., 2019). The authors uncovered cases where the same surrogate outcome was used to validate the drug in subsequent trials, which only increases temporal stability (by prioritizing test-retest reliability) and not the construct validity of the original surrogate outcome (Gyawali et al., 2019).

Although there are valid arguments to justify the use of surrogate outcomes, the system has been fraught with perverse incentives, including allowing for manufactures to gain larger initial market share (Kemp & Prasad, 2017). This has resulted in the burden of evidence to be transferred from clinical trials and onto patient populations, which has

resulted in ineffective, possible unsafe medications being tested on patients (Kemp & Prasad, 2017). A paper that outlines both the case for and against the use of surrogate endpoints is Ciani et al. (2017). Given the demonstrated inappropriate use of surrogate outcomes, there are concerns surrounding the proper application of these pathways, and it is often equivocal as to whether drugs meet the threshold of unmet need that justifies expedited market authorization.

2.4 Safety Concerns with Pimavanserin

Despite pimavanserin being advertised as a novel therapy meeting an unmet need, problems emerged during and shortly after its FDA approval. These primarily come from three sources: 1) the PDAC's dissenting opinions of pimavanserin (Food and Drug Administration, 2016); 2) the ISMP's quarterly report citing concerns (Institute for Safe Medication Practices, 2017); and 3) research stemming from commercial databases and the FAERS. From these sources, two serious concerns have emerged: 1) poor clinical evidence to support its regulatory approval; and 2) increased risk of death, adverse events, and serious adverse events.

2.4.1 Poor Clinical Evidence to Support Regulatory Approvals

Part of the dissenting opinion from PDAC was that pimavanserin received approval relying on one clinical trial (NCT01174004, ACP-103-020) that had a sample size of 199 patients. This decision goes against the guidelines that the FDA set for sponsors back in 1998, which state: "with regard to quantity, it has been FDA's position that Congress generally intended to require at least two adequate and well-controlled studies, each convincing on its own, to establish effectiveness...reliance on only a single study will generally be limited to situations in which a trial has demonstrated a clinically meaningful effect on mortality, irreversible morbidity, or prevention of a disease with potentially serious outcome and confirmation of the result in a second trial would be practically or ethically impossible" (Food and Drug Administration, 1998).

At the time of the FDA's decision to approve pimavanserin's use for patients, two other Phase 3 studies had already been completed on pimavanserin: NCT00477672 / ACP-103-012 (n=298) and NCT00658567 / ACP-103-014 (n=123) by June 2009 and December 2009, respectively. Despite this evidence, these two trials were not considered heavily in the FDA's regulatory decision due to the role of NCT01174004 as the pivotal trial. Based on the aforementioned quote, this decision to primarily consider NCT01174004 / ACP-103-020 also inadvertently and incorrectly categorizes pimavanserin as a drug in which it would be "ethically or practically impossible" to conduct a second trial. This deviation from past FDA guidance raises questions as to the legitimacy and validity of pimavanserin's expedited approval designations.

The reason that the NCT01174004 / ACP-103-020 was heavily weighted in the consideration process is because it is considered the pivotal trial. The FDA does not provide any official guidance documents for the regulatory assessment of these pivotal trials, but an interview from Lexchin et al. (2021) with the FDA yielded this interpretation regarding pivotal trials.

"The term 'pivotal' does not show up in regulations, but its general meaning to people is clear enough: It is the trial or trials that will be, or could be, the basis for our reaching a conclusion that there is 'substantial evidence of effectiveness,' the statutory and regulatory standard for approving a drug. That would also mean that the FDA considers it (or them) an 'adequate and well-controlled investigations,' which is the only basis for accepting a study as supporting effectiveness. A pivotal trial presents the most important data used by FDA to decide whether to approve a drug" (Lexchin et al., 2021).

In the context of pimavanserin's regulatory approval, NCT01174004 / ACP-103-020 acted as the most importance source of data in the FDA's regulatory decision, despite other existing trials suggesting varying degrees of efficacy.

2.4.2 Increased Risk of Death, Adverse Events, and Serious Adverse Events

Given the limited evidence submitted to the FDA as part of the review process, the PDAC expressed their concerns surrounding pimavanserin's risk profile (Food and Drug Administration, 2016). As the sample size was quite small, it would have been hard to ascertain with reasonable certainty whether the deaths, adverse events, and serious adverse events were statistically significant, or if they could be explained by an underlying pathophysiological mechanism.

In November 2017, the ISMP, a non-profit watchdog organization based in the United States dedicated to monitoring drug safety signals, published a quarterly report citing concerning safety signals and adverse reports (Institute for Safe Medication Practices, 2020). These included hallucinations (21.8%; 487/2336), drug ineffectiveness (14.9%; 333/2236), confused state (11.5%; 258/2236), and death (10.9%; 244/2236) (Institute for Safe Medication Practice, 2017).

More recent follow-up studies conducted in 2021 and 2022 using the FAERS as well as commercial insurance databases have indicated that although a safety signal exists, there are no statistically significant differences when compared to existing treatment alternatives like clozapine and quetiapine (Brown et al., 2021; Nguyen et al., 2022). Other research has contested these findings, citing an increased risk of hospitalization at 180 days of utilization and an increased risk of mortality at 90 days, 180 days and 1 year of pimavanserin utilization relative to non-users (Hwang et al., 2021).

Although patients with PD are often older, frail, and have an extensive history of polypharmacy, which could confound potential safety signals, there remain concerns regarding the FDA review process and the clinical trials that led to the pimavanserin's market approval (Hwang et al., 2021). These concerns emerging from both trial data and real-world patient data are potentially problematic because patients could be at increased risk and the onus to protect these patients falls upon the FDA's review processes, which is the focus of this study.

2.5 Relevance to Canadians with Parkinson's Disease

The rigor of pimavanserin's trials should be examined not only to better understand the potential limitations of the FDA's approval processes, but also to inform other markets, such as Canada, where it has not yet been approved. This is timely as patient groups and regulatory agencies are beginning to discuss and track pimavanserin's utilization and pricing in the United States market (Canadian Agency for Drugs and Technologies in Health, 2020; Patented Medicine Prices Review Board, 2018).

In addition, there is also mention of pimavanserin in the official Canadian PD guidelines published in the Canadian Medical Association Journal (CMAJ; Grimes et al., 2019). These guidelines are important documents that help to inform physicians on the most pertinent treatment protocols, scientific evidence, and best practices. The guideline utilized a literature review approach to aggregate evidence in the form of existing guidelines, systematic reviews, and RCTs. The PD guideline specifically mentions pimavanserin as a potential therapeutic of interest, assigning it a "Grade B" recommendation (Grimes et al., 2019). A Grade B recommendation denotes therapies with "a body of evidence including studies rated Level 2++, directly applicable to the target population and demonstrating overall constituency of results" (Grimes et al., 2019). "Level 2++" evidence denotes "High-quality case-control or cohort studies with a very low risk of confounding or bias and a high probability that the relationship is causal" (Grimes et al., 2019). Despite this recommendation, and the growing interest from Canadian stakeholders, the literature appears to suggest that there is still a high degree of uncertainty with pimavanserin's safety and efficacy. Pimavanserin has not yet received market authorization in Canada.

As such, this study aims to critically appraise pimavanserin's clinical trials, and assess the risk of bias and methodological design of the studies. This will ultimately add to the body of evidence that can be used to better ascertain pimavanserin's safety and efficacy profile for patients in markets where pimavanserin has not yet been authorized for use.

Chapter 3

3 Research Questions and Objectives

The chapter will first introduce readers to the importance of methodologically rigorous clinical trials design, after which the research question and objectives will be communicated. This will transition into a justification as to why third-party research reviews like this project, acting independently of the FDA, are warranted. Specifically, Aduhelm, a controversial drug from the manufacture Biogen, will be discussed as an example to support the case for a review of pimavanserin's trials.

3.1 Objective: The Importance of Clinical Trials Design

Clinical trials are critical in determining whether pharmacotherapies are safe and efficacious enough to be prescribed to the public. Poorly executed trials with a high risk of bias may reduce the interpretability of the drug's efficacy, and can potentially omit adverse events, serious adverse events, and mortality risks to regulators. Several pertinent concerns include poor trial design, substandard outcome measurement, inadequate statistical methodology, and missing data (Lexchin, 2011; Little et al., 2012).

As such, this project will be a retrospective study with the research question "was there enough quality evidence and suitably designed trials to warrant an accurate safety and efficacy assessment?" with the primary objective of informing regulatory decisionmaking regarding the use of pimavanserin for the treatment of PDP. By conducting this extensive trial assessment in relation to the standards established by the FDA, the academic and medical community can be better informed about pimavanserin's safety and efficacy profile as an alternative to traditional antipsychotics for the management of PDP.

3.2 Case Example: Biogen & Aduhelm (aducanumab-avwa)

A potential criticism of conducting independent third-party reviews is that the FDA is more qualified to scrutinize trial data and drug applications. While it is true that the FDA is better equipped with more resources to evaluate RCT data, there have been many cases where the FDA has approved new drugs through expedited pathways that should not have been approved. A recent example of this is Biogen's drug Aduhelm, which was indicated for the treatment of Alzheimer's disease, and approved by the FDA on June 7th, 2021, through both the Accelerated Approval and Fast-Track designations. These two designations awarded Biogen the ability to utilize a surrogate outcome and receive FDA support in their application, contingent on demonstrating benefit using nonclinical mechanisms or early pharmacological data.

Despite 10 of the 11 members on the FDA's own Peripheral and Central Nervous System Drugs Advisory Committee voting against Aduhelm, coupled with a lack of scientific support for their surrogate outcome, the FDA approved the drug (Biogen, 2020; Food and Drug Administration, 2021). An independent investigation by STAT news uncovered that the FDA's Director of the Office of Neuroscience, Billy Dunn, allegedly met with Biogen executives in unsanctioned meetings against FDA internal policies, to help push what was otherwise a questionable drug to the forefront of approval (Feuerstein et al., 2022). This is reminiscent of Curtis Wright IV, acting director of the FDA's Division of Anesthetic, Critical Care, and Addiction of Drug Products from 1996 to 1997, who also held unsanctioned meetings with representatives from Purdue Pharma to help push for Oxycontin's approval (U.S. Department of Justice, 2006).

As a reaction to the alleged oversight by the FDA, several retrospective critical appraisals have since been conducted to examine the regulatory processes and clinical evidence that led to Aduhelm's approval. Although each varying in their scope, notable studies include a narrative review by Haddad et al. (2022), a critical appraisal of statistical methods by Knopman et al. (2020), and a critical appraisal of trial design and outcomes by Tampi et al. (2021). The narrative review conducted by Haddad et al. (2022) found that the controversy surrounding Aduhelm was multifaceted and included issues with the FDA's narrow interpretation of the clinical evidence, as well as the lack of statistical significance in some clinical studies, amongst other factors. Upon conducting a critical appraisal of Aduhelm's clinical trials, Both Knopman et al. (2020) and Tamp et al. (2012) reaffirmed the need for additional data by conducting another clinical trial. As a result, it is not uncommon, and arguably of significant value, for independent researchers to conduct retrospective critical appraisals of clinical trials independently of national regulatory authorities, especially in the case of controversial approvals, as each review adds to the body of evidence that can help other regulatory bodies better understand the safety and efficacy of a drug.

Chapter 4

4 Methodology

This study is a retrospective critical appraisal of pimavanserin's published clinical trials, using the Risk of Bias 2 (RoB 2) tool and the PRagmatic Explanatory Continuum Indicator Summary (PRECIS-2). The research consisted of two main stages: 1) collection of clinical trials data; and 2) evaluation using the RoB 2 and PRECIS-2, followed by a discussion that will be situated in the context of the FDA's own decision regarding the licensure of pimavanserin, alongside other extant independent reviews.

4.1 Stage One: Data Collection

The project began by gathering all pertinent clinical trials data from the Clinical Trials database of the US National Library of Medicine from 2007 onward. 2007 was chosen specifically because pimavanserin's approval from the FDA through the Breakthrough Therapy designation was in 2016. A decade prior to FDA approval is sufficient to capture all possible Phase 2 clinical data that Acadia Pharmaceuticals, or another organization, may have reported to the FDA. The two studies uncovered in pimavanserin's patent, both predating 2007, will not be evaluated because they did not contain sufficient information nor a substantial sample size to warrant a critical review of their trials.

There are two classes of clinical trials data specified within this study: the data of 'terminated', 'withdrawn', or 'unknown' status studies; and the data from 'completed'

status studies. The clinical data for both groups was systematically logged using an Excel spreadsheet, recording 1) the trial design type; 2) pimavanserin's dosage or drug delivery mechanism; 3) primary and secondary outcomes; 5) participants' data and/or missing data characteristics; and 6) study conclusions.

4.2 Stage Two: Evaluation Using the RoB 2 and PRECIS-2 Tools

The clinical trials were then evaluated using both the RoB 2 and PRECIS-2 tool. These two tools were used in conjunction because each assesses different aspects of a clinical trial's internal and external validity, resulting in the need for multiple tools, a concern realized by Moher et al. (1995). Both the RoB 2 and the PRECIS-2 tools were chosen because they have been widely used in clinical trials critical appraisals and have been shown to have good reliability and validity (Minozzi et al., 2022; Loudon et al., 2017). By using multiple tools, we were able to adequately capture areas of interest such as statistical methods, prognostic data, internal and external validity, trial design, and overall risk of bias.

It is important to consider that despite checklists and frameworks, there are no definitive means to determine if an RCT is "sound". Even amongst expert statisticians and epidemiologists, there are points of contention as to how the data and trial design should be interpreted and valued; although RCT features like blinding are core tenets, features like prognostic imbalances and sample size may be variably interpreted based on the nature of certain diseases. These competing values often need to be considered in light of existing treatment options, or lack thereof, and patient need. However, just because it is difficult to achieve a resolutely perfect evaluative tool does not mean that the results should be rendered invalid. Likewise, no clinical trial is perfectly designed. Yet, this does not mean that the results of a clinical trial has no interpretative value. Rather, as long as the trial has a robust methodology, it can still build upon the body of clinical evidence. Therein lies the value of the RoB 2, the PRECIS-2, and this study. The results of this study, even if limited in scope, will add to the volume of literature examining the safety and efficacy of pimavanserin, and the robustness of the trials that contribute to its regulatory approval.

4.2.1 Risk of Bias 2 Tool

The Risk of Bias 2 (RoB 2) tool was originally released in 2008 by Cochrane, a not-for-profit organization located in the United Kingdom that aims to support the generation of quality evidence to better inform health decision-making (Cochrane, n.d.). Specifically, the tool is used to assess the methodological quality of randomized clinical trials, and to determine whether a risk of bias could skew the interpretation in favor of the experimental drug. Since the launch of RoB, an updated version called the RoB 2 was released in 2019, which incorporated user feedback and suggestions from the original RoB and has since been applied in a variety of studies (Sterne et al., 2019). The tool being applied to pimavanserin's clinical trials is the updated RoB 2.

The RoB 2 contains five primary domains: 1) bias from the randomization process; 2) bias arising from deviations from the intended intervention; 3) bias from missing outcome data; 4) bias in the outcome measurement; and 5) bias in the reported results. Altogether, the risk of bias from each of the five domains contribute to one of

three overall risk of bias scores: 1) low risk of bias; 2) some concerns; or 3) high risk of bias. An overall "low risk of bias" score is achieved only when each of the five domains have a "low risk of bias" score. An overall "some concerns" score is achieved if at least one of the five domains has a "some concerns" score, but no "high risk of bias" score for any domain. An overall "high risk of bias" score is achieved if any one of the five domains has a "high risk of bias" score, or if multiple domains have a "some concerns" score.

The RoB 2 tool can be applied in one of two ways, either manually or autonomically using a macro-enabled Excel tool. For the first option, users apply the prescribed algorithms based on a set of domain-specific signaling questions created and validated by Cochrane to determine the domain-specific risk of bias. An algorithm in this tool refers to the set of rules and pathways processes that determine the risk of bias rating. An example algorithm from domain 1 (bias) from the randomization process has been included as *Figure 6* to illustrate its application. Alternatively, a user can employ an



Figure 6: Algorithm for suggested judgement of risk of bias arising from the randomization process (Sterne et al., 2019).

Excel document with macros enabled which automatically apply the algorithms based on user-provided answers to the signaling questions. A complete list of the signaling questions from each of the given domains have been included in *Table 3*. To reduce the potential for misinterpretation and to streamline the process, the macro-enabled Excel tool was used in this step of the study.

Domain	Signaling Question			
1. Risk of Bias Arising	-Was the allocation sequence random?			
from the Randomization Process	-Was the allocation sequence concealed until participants were enrolled and assigned to interventions?			
	-Did baseline differences between intervention group suggest a problem with the randomization process?			
2. Risk of Bias Due to	-Were participants aware of their assigned intervention during the trial?			
Deviations from the Intended Interventions	-Were carers and people delivering the intervention aware of participants' assigned intervention during the trial?			
	-Were there deviations from the intended that arose because of the trial context?			
	-Were these deviations likely to have affected the outcome?			
	-Were these deviations from intended intervention balanced between groups?			
	-Was an appropriate analysis used to estimate the effect of assignment to intervention?			
	-Was there potential for a substantial impact (on the result) of the failure to analyze participants in the group to which they were randomized?			
3. Risk of Bias Due to	-Were data for this outcome available for all, or nearly all, participants randomized?			
Missing Outcome Data	-Is there evidence that the result was not biased by missing outcome data?			
	-Could missingness in the outcome depend on its true value?			
	-Is it likely that missingness in the outcome depended on its true value?			
4. Risk of Bias in	-Was the method of measuring the outcome inappropriate?			
Measurement of Outcome	-Could measurement or ascertainment of the outcome have differed between intervention groups?			

	-Were outcome assessors aware of the intervention received by study participants?		
	-Could assessment of the outcome have been influenced by knowledge of intervention received?		
	-Is it likely that assessment of the outcome was influenced by knowledge of intervention		
	received?		
5. Risk of Bias in	-Were the data that produced this result analyzed in accordance with a pre-specified analysis		
Selection of the Reported	plan that was finalized before unblinded outcome data were available for analysis?		
Result	-Is the numerical result being assessed likely to have been selected, on the basis of the results, from multiple eligible outcome measurements (e.g. scales, definitions, time points) within the		
	outcome domain?		
	-Is the numerical result being assessed likely to have been selected, on the basis of the results, from multiple eligible analyses of the data?		

 Table 3: Complete list of signaling questions from the RoB 2 tool.

It should be noted only four of eight studies gathered were randomized and blinded trials. Three of the four remaining studies were single arm open-label studies. One of the four remaining studies was an open-label extension study of another RCT. The descriptive breakdown of the studies, their phases, and trials designs, are detailed under the 'Results' section.

Even though the RoB 2 tool is meant to be applied to randomized trials, the three single arm assignment trials were also assessed with the RoB 2 because certain signaling questions within domain 3 (missing data), domain 4 (measurement of the outcome), and domain 5 (selection of the report results) can still be applied. The domains that are applicable to the single arm studies, along with their limitations, are discussed further under the 'Results' and 'Discussion' sections. It should also be acknowledged that single arm studies have an important role and add value to clinical trials but should be considered in light of their risk of bias and statistical limitations (Cucherat et al., 2020). At the moment, there are no widely validated tools for the assessment of bias for single arm studies, so the RoB 2 was employed on all of the studies, RCT or otherwise.

Using the RoB 2 tool on single arm studies is not a completely novel approach. Checcucci et al. (2021) employed the original RoB tool, which is also designed for randomized trials, to review single arm studies investigating ultra-minimally invasive surgical treatments on perioperative and functional outcomes. As well, Gupta et al. (2016) employed the RoB tool to evaluate single arm studies investigating the effects of positive airway pressure on depression in patients with obstructive sleep apnea. As such, employing the RoB 2 tool in this study in the absence of a validated tool meant for single arm trials, is a similar approach adopted by other investigators.

4.2.2 PRECIS-2 Tool

The PRagmatic Explanatory Continuum Indicatory Summary (PRECIS) is a tool that can be used to evaluate whether a clinical trial is more pragmatic or explanatory in nature (Loudon et al., 2015). Originally developed from 2005-2008, the tool can be applied prospectively by clinical trialists to design a trial suitable for their needs or, as with this study, applied retrospectively to examine whether a clinical trial is "fit for purpose" (Loudon et al., 2015). Since its release, it has been updated based on user feedback to include a better scoring system, reduce redundant domains, amongst other improvements, to yield the improved PRECIS-2. The tool being applied to pimavanserin's clinical trials is the updated PRECIS-2 tool.

The purpose of this tool is to provide insight into nine domains, which include: 1) eligibility; 2) recruitment; 3) setting of the trial; 4) organization of trial resources; 5) flexibility with the intervention delivery; 6) flexibility with intervention adherence; 7) the degree of participant follow-up; 8) relevance of the primary outcome; and 9) the comprehensiveness of the primary analysis (Loudon et al., 2015). Each domain is rated from one, representing very explanatory, to five, representing very pragmatic. Domains that are not applicable are not ranked.

A highly pragmatic study is one that is designed to emulate real world conditions with the usual standard of care, and with heterogenous patient adherence levels, amongst other factors (Treweek & Zwarenstein, 2009). A highly explanatory trial is one that is designed and carried out to confirm a casual hypothesis between a therapeutic intervention and a physiological outcome under stringent testing conditions on a specific population (Treweek & Zwarenstein, 2009). It is important to keep in mind that a study does not need to be completely pragmatic or explanatory, as these classifications exist on a continuum rather than on a binary scale (Loudon et al., 2015). A PRECIS-2 has been included as *Figure 7* to illustrate how a completed exampled would be presented. This example illustrates a highly pragmatic trial. Even within the nine different domains, certain design features may be more pragmatic, while others are more explanatory, as illustrated by the example in *Figure 7*.



Figure 7: Example of a completed PRECIS-2 tool (Aronson et al., 2008).

Neither pragmatic study designs nor explanatory trial designs are inherently 'better' than the other; each have their own respective applications and should be judged for suitability based on the objectives of the study. In the context of pimavanserin's drug trials or trials for other psychiatric drugs, an explanatory study might be best suited for determining the efficacy of the drug, while a pragmatic trial might be more applicable for a study investigating its safety profile (e.g., mortality risk, adverse events), especially with concerns surrounding polypharmacy in this patient population (McLean et al., 2017). The application of the PRECIS-2 on the aggregated pimavanserin trials will help to illustrate the interpretability of the trial data in the context of its primary outcome, whether it is efficacy or drug safety.

Chapter 5

5 Results

The results will be presented to first illustrate findings from the data collection process. This will be followed by a description of the eligible studies, including their study designs, length of trials, amongst other descriptive parameters. Finally, the specific primary and secondary endpoints, along with their suitability, will be presented.

The results of the Risk of Bias 2 (RoB 2) tool will then be presented. High-level summary figures will be presented to illustrate the areas that present the greatest risk of bias. This will be followed by a presentation of the most prominent risk of bias that were uncovered using the RoB 2 tool. The presentation of these results will be grouped based on their respective study designs, either as an RCT, a safety study, or a single arm study. This is because each type of study serves a different purpose in determining the safety and efficacy of the drug, and because certain study designs (i.e., RCTs), should be weighed more heavily than other study designs (e.g., single arm studies).

Finally, the results of the PRECIS-2 tool for each will be presented in a graphical format to illustrate whether designs were more explanatory or pragmatic. Similar to the results of the RoB 2 tools, the studies evaluated with the PRECIS-2 tool will be presented based on their respective study designs, either as an RCT, a safety study, or a single arm study.

5.1 Description of the Collected Studies

5.1.1 Eligible Studies

A total of 14 studies were identified through the clinicaltrials.gov search An adapted PRISMA flow chart has been included as *Figure 8* (Page et al., 2021). Inclusion criteria included studies that were sponsored by Acadia Pharmaceuticals, were investigating patients with PD, and had relevant endpoints related to psychosis or general psychiatric functioning. Of those 14 studies, two studies were excluded because their results had not been reported at the time of the FDA decision. As such, these two studies would not provide any pertinent information about the FDA's original decision to approve pimavanserin. Of the remaining 12 clinical studies initiated by the sponsor, three were excluded because the study had been terminated, their status was unknown, or they were still recruiting. One other study was removed because it was part of the sponsor's application for dementia-related psychosis and not part of the PD patient population. As such, these studies did not contribute any data to the regulatory decision-making process.

One study (NCT03482882), titled an "An Open-Label, 8-Week Study of Safety and Efficacy of Pimavanserin Treatment in Adults with Parkinson's Disease and Depression" was included despite not being specifically meant for PDP. It was included because it was conducted on patients with PD and had relevant secondary endpoints, including the Clinical Global Impression-Improvement (CGI-I) at week 8 and Clinical Global Impression-Severity (CGI-S) at week 8. Both secondary endpoints are global psychiatric functioning measures used to assess overall patient health.



Figure 8: An adapted PRISMA diagram of eligible studies.

5.1.2 Description of the Eligible Studies

Of the eight studies that were completed and funded by Acadia Pharmaceuticals, there were three Phase 2, four Phase 3, and one Phase 4 study. Four of the studies were parallel assignment RCTs (i.e., they have a comparator group), while three were single group assignment studies with no control arm.

One of the eight studies (NCT00550238) was a prospective cohort study where participants were selected by investigators from completed pimavanserin RCTs. These participants were selectively chosen because the investigators believed they would benefit from further treatment with pimavanserin. These study participants were categorized as either receiving concomitant antipsychotic medications which was mainly quetiapine and clozapine, or not receiving concomitant antipsychotic medications. Although all participants from each group received pimavanserin, the concurrent antipsychotic medications were prescribed by clinicians outside the investigation team and were considered as the 'exposure'. As such, this study can be best described as a prospective cohort study, where the outcomes are mortality, serious adverse events, and adverse events, and the exposure being other atypical antipsychotics.

Five of the eight trials had intervention periods between 6-8 weeks, inclusive of follow-up time. The sole Phase 4 open-label trial had a primary endpoint of 16 weeks. Two of the open-label safety studies investigating mortality, serious adverse events, and adverse events, had indefinite endpoints. This meant that patients were considered enrolled in the study until patients ceased treatment because of an adverse event, death,

irremediable disease progression, voluntary withdrawal by subject, physician decision, poor efficacy, patient non-compliance, lost to follow-up, or withdrawn by the sponsor.

The comprehensiveness of the data varied among the studies, meaning that some studies only published their data results on clinicaltrials.gov, whereas other studies also published their protocols and analysis in journal articles. Optimally, every study would publish their results on clinicaltrials.gov, as well as their protocol, statical analysis plan, and a journal article. Unfortunately, this poor reporting of potentially negative results, termed 'positive publication bias', is a chronic issue within the scientific community (Mlinarić et al., 2017). The degree to which data results, journal articles, and protocols were reported was an important factor to capture the scope of each study, and substandard reporting of these sources can make it difficult to capture the risk of bias as well as ascertain the drug's safety and efficacy profile.

Most of the trials (n=6) posted their study results on clinicaltrials.gov, as required by the FDA Amendments Act (FDAAA) of 2007. Two studies (NCT00087542, NCT04292223) that had recently concluded were within the 12-month grace period allowed by the FDAAA, meaning they were not yet required to post their results publicly as of the conclusion of this project. Three clinical trials were posted with journal articles and two were also posted with the statistical analysis plan and study protocol. Three trials were posted with only their study results on clinicaltrials.gov. A summary of these study components has been included in *Table 4* and a detailed summary of the studies can be found in *Supplementary File A*.

Study Design		Percentage of Total Studies
RCT (Parallel Assignment)	4	50%
Uncontrolled Single Group Assignment	3	37.5%
Prospective Cohort Study	1	12.5%
Length of Trial		
6-8 Weeks	5	62.5%
16 Weeks	1	12.5%
Indefinite	2	25%
Comprehensiveness of Data Results		
1) Trial with a Study Results Posted	6	75%
2) Trials with a Journal Publication	3	37.5%
3) Trials with a Posted Statistical Analysis Plan	2	25%
4) Trials with a Posted Study Protocol	2	25%
Trials with 1), 2), 3), and 4)	2	25%
Studies with only 1)	3	37.5%

Table 4: Summary of study components.

5.1.3 Primary and Secondary Outcomes and Scales Utilized

Due to the multifaceted nature of PDP and its associated management, there were a variety of primary and secondary outcomes and tools employed within the clinical trials. They can be broadly categorized into five groups based on their relation to a patient's care: 1) psychosis specific endpoints; 2) general psychiatric endpoints; 3) other PD endpoints (e.g., motor function, sleep); 4) care partner endpoints; and 5) safety endpoints. They are summarized in *Table 5*.
Type of Endpoint	Name of Endpoint / Tool	Number of Times Used (n)
Psychosis Specific Endpoint	SAPS-PD (Scale for the Assessment of Positive Symptoms)	3
General Psychiatric	CGI-I (Clinical Global Impression-Improvement)	3
	CGI-S (Clinical Global Impression-Severity)	3
	PGI-I (Patient Global Impression of Improvement)	1
Other PD Endpoints	UPDRS-2 & 3 (Activities of Daily Living, Motor Impairment)	3
	MDS-UPDRS Part 1 and 2 (Movement Disorders Society - Unified PD Rating Scale - Caregiver and Patient Version)	1
	Schwab and England ADL Scale (Caregiver and Patient Version)	1
	SCOPA-SLEEP (Scale for Outcomes in PD-Sleep Scale)	2
	mFSQ (Modified Functional Status Questionnaire)	2
	EQ-5D	1
Caregiver Related	(CBS) Zarit 22 Item Caregiver Burden scale	1
Safety Endpoints	Mortality, Serious Adverse Events, Adverse Events	2

Table 5: different endpoints used in the clinical trials, along with their frequency of

use.

The primary tool used to assess symptoms of psychosis was a modified version of the Scale for the Assessment of Positive Symptoms (SAPS), which is administered using a formal clinical interview between a clinician and a patient (Fernandez et al., 2008). The PD-specific version, the SAPS-PD, is a shortened version of the original SAPS that includes only the subsections assessing hallucinations and delusions and was used three times throughout the trials (Voss et al., 2012). The original SAPS was a recommended tool by the Movement Disorder Society Task Force of the International Parkinson and Movement Disorder Society (Fernandez et al., 2008). Although the SAPS-PD shows strong validity and reliability, it should be noted that research validating the SAPS-PD was conducted by investigators from Acadia Pharmaceuticals, the drug company behind pimavanserin (Voss et al., 2013).

Another psychiatric scale, the Clinical Global Impressions Scale (CGI), which is not psychosis specific, was also employed within the clinical studies. Within the CGI are two individual scales, the severity scale (CGI-S) and the improvement scale (CGI-I), each with only one signaling question. The former assesses the severity of the mental illness based on the patient's function, symptoms, and behaviors while the latter is used to assess the patient's improvement after initiating a new therapy (Busner & Targum, 2007). Both can be used to assess a patient's disease course over time and are commonly used in clinical trials that are part of FDA submissions (Busner & Targum, 2007). The CGI-I and CGI-S were each employed three times throughout the trials.

The Unified Parkinson's Disease Rating Scale (UPDRS), developed in 1987, was a prominent assessment tool for patients with PD. It contains four scales: 1) mentation, behavior, and mood; 2) activities of daily living (ADL); 3) motor; and 4) complications. In 2008, the Movement Disorder Society (MDS) conducted revisions to improve the UPDRS to yield the MDS-UPDRS (Goetz et al., 2008). The revised tool can be completed by either the physician, the patient, or a combination of the two, and contains four scales: 1) non-motor experiences of daily living; 2) motor experiences of daily living; 3) motor examination; and 4) motor complications (Goetz et al., 2008). Although the UPDRS and MDS-UPDRS both assess hallucinations and delusion to some degree, their focus is on PD as a whole, and as such, were not categorized as a psychosis specific or as a general psychiatric tool. The Schwab and England Activities of Daily Living Scale was another tool that was used by the clinical trials to assess a patient's degree of independence and capacity for completing their ADLs.

Other tools used to assess non-psychiatric aspects of PD included the Scale for Outcomes in PD Sleep Scale (SCOPA-SLEEP), the Modified Functional Status Questionnaire (mFSQ), and the EuroQol-5-Dimensions-5 Levels (EQ-5D-5L). The SCOPA-SLEEP is a patient-administered tool that facilitates a clinician's understanding of their nighttime sleep and daytime sleepiness (Marinus et al., 2003). This is of particular interest because the etiology of sleep issues experienced by PD patients are often poorly understood and, as a result, inadequately managed (Adler & Thorpy, 2005). The mFSQ is a general patient-administered assessment tool that captures aspects of the patient's physical function as it related to ADL, psychological function, role function, social function, and other performance measures (e.g., work, sexual relationships, etc.) (Jette et al., 1986). The EQ-5D-5L is a quality-of-life tool which contains five domains: 1) mobility; 2) self-care; 3) usual activities; 4) pain/discomfort; and 5) anxiety/depression (Williams, 1990). The '5L' refers to the five levels that patients can rate the severity of their problems (Williams, 1990).

Another important aspect of PD's management that was assessed was the wellbeing of care partners. Caring for someone with PD is usually taxing for care partners, who are commonly close family and loved ones, and caregiver burden is often worsened by psychiatric symptoms, falls, and the patient's own decreasing quality of life (Schrag et al., 2006). In turn, care partners themselves experience worsening depression and dissatisfaction with their marriage, social, and sexual life (Schrag et al., 2006). The only relevant tool employed in pimavanserin's clinical trials was the Zarit 22 Item Caregiver Burden Scale. Originally developed in 1980 by Zarit et al.. The version used in the clinical trials is a reduced 22 item version that aims to elicit care partner perspectives surrounding the dependence of the individual with PD, as well as their own feelings of burnout, frustration and anger, the strain on their own lives, and other dimensions (Zarit et al., 1980).

Finally, for the two safety studies, the outcomes measures were counts of mortality, adverse events, and serious adverse events.

5.2 Risk of Bias 2 (RoB 2)

The RoB 2 was applied to all eight of the reviewed clinical trials. As mentioned previously, there are five domains that together yield the overall risk of bias for each study. These domains are as follows: 1) Randomization process; 2) Deviations from the Intended Interventions; 3) Missing Outcome Data; 4) Measurement of the Outcome; and 5) Selection of the Reported Result. *Figure 9* illustrates the domain-specific risk of bias and overall risk of bias for each study.



Figure 9: RoB 2 summary table.

Overall, there were methodological concerns with every study conducted by the sponsor. The most prominent issue that was uncovered was the poor management of missing data, with six of eight studies indicating major risk of bias from this issue. Another prominent issue was deviations from the intended intervention, with six of eight studies indicating major risk of bias. *Figure 10* has been included to illustrate the risk of bias as a percentage of the total studies, and the domains that produced the greatest risk of

bias within these studies. *Supplementary File B*, the RoB 2 Excel document with the original ratings and rationales, has also been included.



Risk of Bias as Percentage

Figure 10: Risk of bias as a percentage of total bias.

As previously mentioned, the following sub-sections present the methodological concerns identified for each type of clinical trial, whether they were an RCT, a safety study, or a single arm study. The reason the results are presented in this manner is because the RoB 2 is best applied to RCTs, and as such, should be presented separately from the other types of study designs for which only certain parts of the RoB 2 are applicable. As well, each study design has their own methodological concerns, and should be presented and discussed separately. The results in the following paragraphs will also be presented with contextual information gathered from their respective journal articles, statistical analysis protocols, study protocols, amongst other applicable resources.

5.2.1 Randomized Control Trials

5.2.1.1 Randomization

There were a total of four RCTs (NCT00087542, NCT01174004, NCT00477672, and NCT00658567). One of those trials (NCT00087542) did not have any results posted as it was completed in 2005 before mandatory reporting was in effect, so it was impossible to ascertain issues with randomization, deviations from the intended intervention, or issues with the reporting of the results.

However, amongst the remaining three RCTs, the randomization process contributed to a low risk of bias as there were robust methodologies in place. For example, NCT01174004 used a pre-programmed kit randomization schedule generated by PharamaNet, ensuring robust randomization and allocation blinding. The other studies, however, did not publish their specific protocols, but can reasonably be assumed to have randomization as per Cochrane's user protocol (i.e., cribsheet), and as a result, were rated as having a low risk of bias.

5.2.1.2 Missing Outcome Data

A major source of bias that contributed to poor RoB 2 ratings within the RCTs was unexplained or poorly explained missing outcome data. In NCT01174004, the landmark RCT that formed the basis of pimavanserin's approval, 23 out of 199 participants (11.56%) withdrew from the study and data was not reported on them: 7 out of 94 from the placebo arm and 16 out of 105 from the experimental arm. The reasons cited were vague and broadly categorized as either removal due to an adverse event,

withdrawal by subject, sponsor's discretion, protocol violation, or physician decision. Missing data stemming from adverse events may indicate safety issues, and of even greater concern are participants withdrawn due to sponsor's discretion and physician decision, which may point to questionable management of efficacy data. No statistical compensatory methods were employed. Resulting from the sponsor's poor explanation as to the nature of the absence of the data besides broad generalizations, and due to their poor statistical methods for handling missing data, this study was rated as having a high risk of bias, which may have contributed to the positive efficacy conclusion drawn by Acadia Pharmaceuticals. *Table 6* illustrates the specific breakdown of missing outcome data from each RCT.

The issue of poorly explained missing data is not unique to NCT01174004, as both NCT00477672 and NCT00658567 presented with a similar issue. With NCT00477672, consent was withdrawn by 15 patients across the 10mg (total n = 101) and 40mg (total n = 99) treatment arms, the reasons for which have not been provided by the sponsor. Since there were no issues with prognostic baseline imbalances and the treatment arms were similar in size, there may be reason to believe that the patients who withdrew consent from participation in the treatment arm points to greater safety risk that has not yet been acknowledged by the sponsor.

Study (NCT01174004)	Control Arm	Pimavanserin Arm	
Total n	94	105	
Missing n	7	16	
Percentage of Drop-Out	7.45%	15.24%	
Adverse event	2	10	
Withdrawal by Subject	2	3	
Sponsor Discretion	2	2	
Protocol Violation	0	1	
Physician Decision	1	0	
Study (NCT00477672)	Control	Pimavanserin Arm 40mg	Pimavanserin Arm 10mg
Total n	98	101	99
			<i>))</i>
Missing n	7	16	16
Missing n Percentage of Drop-Out	7 7.14%	16	16 16.16%
Missing n Percentage of Drop-Out Adverse Event	7 7.14% 3	16 15.84% 6	16 16.16% 5
Missing n Percentage of Drop-Out Adverse Event Death	7 7.14% 3 0	16 15.84% 6 0	16 16.16% 5 1
Missing n Percentage of Drop-Out Adverse Event Death Disease Progression	7 7.14% 3 0 0	16 15.84% 6 0 0	16 16.16% 5 1 1
Missing n Percentage of Drop-Out Adverse Event Death Disease Progression Physician Decision	7 7.14% 3 0 0 1	16 15.84% 6 0 0 0	16 16.16% 5 1 1 0

Consent Withdrawn	2	10	5
Sponsor Discretion	1	0	2
Study (NCT00658567)	Control	Pimavanserin Arm 20mg	Pimavanserin Arm 10mg
Total n	40	42	41
Missing n	8	6	4
Percentage of Drop-Out	20%	14.29%	9.76%
Adverse Event	5	3	2
Consent Withdrawn	2	2	0
Physician Decision	0	0	1
Sponsor Discretion	1	1	1

Table 6: Breakdown of missing data from each treatment arm.

5.2.1.3 Deviations from Intended Intervention and Selection of the Reported Result

Building upon the issue of missing data, there were major deviations from the intended intervention in the RCTs. For NCT00658567, the study was prematurely arrested and the statistical analysis for the 10mg was not conducted. The sponsor did not provide a reason for this deviation. As a result, this failure to implement the intervention led to missing data for both the primary and secondary outcomes in this treatment arm. This also led to selective reporting and analysis of the primary and secondary endpoint data for the only remaining 40mg treatment arm.

5.2.1.4 Measurement of the Outcome

Of the four RCTs, three received a low risk of bias rating and one received a moderate risk of bias rating for this domain. In addition, the studies were blinded and randomized, so patients were less likely to provide biased outcome data. These precautions also helped to minimize the risk of bias arising from a clinician's assessment of the outcome. As well, the endpoints and tools that were previously discussed as being used to measure the outcome were appropriate for the scope and objective of the trials. This is because the RCTs employed a validated psychosis specific tool (i.e., the SAPS-PD) alongside other well-established PD and quality of life tools. As such, there were limited concerns regarding the measurement of the outcome across the RCTs.

5.2.2 Safety Studies

There were two safety studies (NCT01518309 and NCT00550238) and both were open-labelled extension studies. Since the RoB 2 is intended for RCTs, the following paragraphs will present methodological issues from select domains that apply to these studies. These domains were chosen because of their potential risk of bias, and because of their relevance to the interpretation of pimavanserin's safety profile. As well, methodological issues that are not adequately captured by the RoB 2 signaling questions will also be presented alongside their most relevant domain.

5.2.2.1 Randomization Process and Participant Selection

As these were non-randomized studies, there is a risk of bias that could confound the results of the study. The lack of randomization is not, however, the primary concern as non-randomized trials are not unusual (Singh & Loke, 2012). Rather, the concern is how patients were initially recruited. Patients for both safety trials originated from existing pimavanserin RCTs and were selected to participate by the investigators based on their perceived benefit with the continued use of pimavanserin. For NCT01518309, the sponsor selected patients "who had completed study ACP-103-006 (PD psychosis [PDP]) or study ACP-103-004 (PD with dyskinesias) and would benefit from continued pimavanserin treatment, as judged by the investigator" (U.S. National Library of Medicine, 2020b). For NCT00550238, the sponsor selected patients that "were adult males or females with Parkinson's Disease Psychosis who had completed a prior doubleblind study with pimavanserin and were determined by the treating physician (i.e. the investigator) to benefit from continued treatment" (U.S. National Library of Medicine, 2019).

This choice of participant selection presents a large risk of bias as it may allow patients who are benefiting from pimavanserin or are experiencing few or no adverse events to selectively participate in the study.

5.2.2.2 Measurement of the Outcome

The design of the studies also poses a risk of bias. For NCT01518309, the statistical analysis of mortality, adverse events, and serious adverse events was calculated for the single group exposed to varying doses of pimavanserin, ranging from 20mg to 60mg. The results for risk were not calculated with consideration for the level of exposure different patients had experienced. As such, patients who were treated with 20mg were analyzed together with patients treated with 40mg or even 60mg of pimavanserin. Without stratification for dosage in the analysis, if a risk were to exist at higher doses, it may be averaged out by participant data from lower dosages. Alternatively, if a risk were to exist at a lower dose, it may be averaged out by participant data from higher doses.

For NCT00550238, the IRR for mortality, adverse events, and serious adverse events were calculated separately for two groups: one group for patients who were treated with pimavanserin 40mg concurrently with other antipsychotics and the other group for patients treated with only pimavanserin 40mg. The most pressing issue with NCT00550238 is that the study did not actually ascertain the standalone risk of pimavanserin, but rather ascertained the risk of concurrent atypical antipsychotic use, which lends limited insight into the safety of pimavanserin. For both safety studies, a better methodological design would be to have two treatment arms, one for the treatment with pimavanserin and the other with either placebo or existing atypical antipsychotics. In the FDA's own guidance document for sponsors, the instructions for the design of safety studies should include a suitable comparator. Specifically, "the ability of a comparative trial to detect a difference between treatments when one exists needs to be established because a trial incapable of distinguishing between treatments that are in fact different cannot provide useful comparative information" (Food and Drug Administration, 2001).

Employing an appropriate comparator group, either placebo or another atypical antipsychotic would have allowed the risk profile of pimavanserin to be more accurately compared and understood.

5.2.3 Single Arm Studies

There were two single arm studies (NCT03482882 and NCT04292223). Again, as the RoB 2 was not designed to evaluate single-arm studies, only applicable methodological issues in select domains will be presented. Only the results for NCT03482882 will be discussed as the results of NCT04292223 were not posted on clinicaltrials.gov because it was only recently completed and still falls within the 1-year grace period of mandatory reporting.

5.2.3.1 Deviations from the Intended Intervention and Measurement of the Outcome

As with the safety studies, there is a substantial risk of bias associated with an unblinded, non-placebo-controlled study. This design may have resulted in both the investigators and study participants being influenced throughout the delivery of the intervention and may have also introduced bias in the measurement of the endpoints. For NCT03482882 specifically, the clinician investigator may have been biased when assessing the severity and clinical improvement of the patient while administering the CGI-S and CGI-I. These issues contributed to the "high risk of bias" ratings for both domain 2 (deviations from intended intervention) and domain 3 (measurement of the outcome).

5.2.3.2 Missing Outcome Data

For NCT03482882, there was an issue with missing data, which may have been a source of bias. From a total participant size of 47, seven participants (14.89%) did not complete the trial, resulting in a concerning degree of missing data. As with the previously discussed RCTs, there was an inadequate explanation as to the source of the patient attrition, beyond vague and broad categories. These have been illustrated in *Table 7*.

Study (NCT03482882)	Pimavanserin 34mg (n=7)
Total n	47
Percentage of Drop-Out	14.89%
Adverse Event	3
Lost to Follow-Up	1
Protocol Violation	2
No Further Specifications	1

Table 7: Breakdown of missing data from NCT03482882.

There was a statistical analysis protocol that explained how missing data were handled and the compensatory methods employed. The investigators utilized different compensatory methods based on the endpoints being valued. For the CGI-S, CGI-I and EQ-5D-5L, missing values were not inputted. For SCOPA-DS and SCOPA-NS, mean imputation was used to compensate for missing values. Overall, due to serious shortcoming in the description regarding the categorization of the missing data, the domain was rated as being at a "high risk of bias".

5.3 PRECIS-2

The PRECIS-2 was applied to all eight trials. The detailed ratings and rationales can be found in *Supplementary File C*. It should be noted that due to the limited availability of published journal articles (n=3), study protocols (n=2), and statistical

analysis plans (n=2) for some studies, certain domains for the PRECIS-2 are necessarily rated as "not applicable".

5.3.1 Randomized Control Trials

For the four RCTs, their study design was mainly explanatory in nature. Across all the domains, the average score amongst the RCTs was 2.139, with the greatest explanatory domains being follow-up (score 1.00), eligibility (score 1.75), and primary analysis (score 2.00). The specific domain scores can be found in *Supplementary File C* and the PRECIS-2 graphs can be found below if *Figure 11*.



Figure 11: PRECIS-2 chart for RCTs.

5.3.1.1 Follow-Up

The follow-up was rated as being highly explanatory because there were frequent check-ins for measurement of the primary and secondary endpoints. This intensity of check-ins is much greater than that of usual care, leading to a low rating. Another contributor to the explanatory rating was that the overall follow up period was quite short, between 6-8 weeks, and does not represent the longitudinal nature of PDP, which can be experienced by patients for years (Friedman, 2010). As such, the initial intensity of visits and the subsequent limited follow-up period caused this domain to be assessed as highly explanatory.

5.3.1.2 Eligibility

The eligibility criteria for the RCTs were explanatory, with strict inclusion and exclusion criteria for participants. The inclusion criteria for the studies required the patients have at least three cardinal features of PD (e.g., rest tremor, rigidity, bradykinesia), psychosis (e.g., visual and/or auditory hallucinations) lasting four weeks or longer, clinical validated psychosis, be on stable doses of PD medication, a reliable caretaker, amongst other stringent inclusionary criteria. Exclusionary criteria included people who were pregnant or breastfeeding, had systematic factors that could contribute to psychosis, history of significant pre-morbid psychiatric conditions (e.g., depression, etc.), dementia precluding accurate assessment, use of depot neuroleptic in the past year, prior exposure to non-depot neuroleptic except for quetiapine or clozapine, amongst other highly stringent exclusionary factors.

5.3.1.3 Primary Analysis

Another contributor to the explanatory nature of the RCTs was how the primary analysis was conducted with regards to missing data. One of the studies (NCT01174004) simply did not impute any of the data, which is a highly explanatory means to handle missing data. Two of the studies (NCT00477672 and NCT00658567) employed a last observation carried forward (LOCF) method, which imputes missing data using the last collected data value and is a more pragmatic approach to missing data than simply not imputing the data. However, for NCT00658567, an entire arm was terminated prematurely with no reason cited, even though data were collected, which led to a highly explanatory rating. For NCT00087542, no statistical analysis plan was provided, so no rating for primary analysis could be provided.

5.3.2 Safety Studies

The two safety studies (NCT01518309 and NCT00550238) were mainly pragmatic in nature. Their PRECIS-2 diagrams can be found below as *Figure 12* and detailed rationales and ratings can be found in *Supplementary File C*. The average PRECIS-2 score across all domains were 4.032. The greatest contributors to this highly explanatory score were eligibility (score 4.00), setting (score 5.00), follow-up (score 5.00), and primary outcome (score 5.00).



Figure 12: PRECIS-2 chart for safety studies.

5.3.2.1 Eligibility

In both studies, patients were selected from patients participating in existing RCTs based on a physician's determination that they would benefit from continued treatment with pimavanserin. Although the RCTs from which the patients were drawn had strict eligibility criteria and could lead to the studies being assessed as more explanatory in nature, the fact that there was a clinical decision applied for patient eligibility led to a pragmatic score. This is because in usual care, a physician would make a similar clinical decision where only patients deemed to likely benefit from a drug would be prescribed that drug.

5.3.2.2 Setting

The setting for these studies was in an outpatient setting. This means that patients were not in a highly controlled clinical environment for the duration of the study, and could continue their treatment with other PD medications, and in the case of NCT00550238, other concurrent atypical antipsychotics. This study design feature is

congruent with how long-term safety studies are normally conducted, and emulated realworld interactions that patients may be in. As a result, this domain for safety studies was highly pragmatic in nature.

5.3.2.3 Follow-Up

As a result of the outpatient setting, patients were observed until they encountered an adverse event, a serious adverse event, or death. This length of follow-up varied and in NCT00550238, patients on concurrent atypical antipsychotics were followed for an average of 172 days while patients not on concurrent atypical antipsychotics were followed on average for 421 days. This domain of the study design was highly pragmatic in nature due to the extensive length of follow-up.

5.3.2.4 Primary Outcome

The primary outcomes for the safety studies were adverse events, serious adverse events, or mortality. These outcomes are of obvious importance to patients, unlike more explanatory outcomes such as surrogate or physiological outcomes. As a result, the primary outcome domain for the two safety studies were rated as highly pragmatic.

5.3.3 Single Arm Studies

The two single-arm studies (NCT04292223 and NCT03482882) had an average score of 2.7, leaning more towards pragmatic than explanatory. The results of these domains will not be extensively discussed because of limited information stemming from the lack of published protocols and statistical analysis plans. For both studies, four of the nine domains did not have sufficient information to rate their domains. Their PRECIS-2 diagrams can be found below as *Figure 13* and their detailed rationales and ratings can be found as *Supplementary File C*.



Figure 13: PRECIS-2 chart for single-arm studies.

5.3.3.1 Eligibility

The inclusion and exclusion criteria for both studies were quite extensive but less stringent than the RCTs. For both, patients had to be diagnosed with PD and had psychotic symptoms that developed after the onset of PD, along with being stable on PD medication.

5.3.3.2 Primary Outcome

Both primary outcomes were global performance scales that are highly applicable to patients, and as a result, were more pragmatic. For NCT03482882, the outcomes used were the CGI-S and the CGI-I, which gauge the patients' disease severity and degree of improvement after initiating treatment. Both endpoints are pragmatic because they relate directly to a patient's day-to-day health and can be applied in routine care (Loudon et al., 2015). This contrasts with explanatory outcomes, which are more esoteric in nature, and may not be of obvious importance to a patient. Examples of explanatory outcomes can include surrogate endpoints and physiological outcomes (Loudon et al., 2015). For NCT04292223, the primary outcome was the mSFQ, which assessed ADL, psychological function and mental health, and social function (e.g., work, social activity, quality interaction, sexual life, etc.), which led to a pragmatic rating.

Chapter 6

6 Discussion

In light of the breadth of information presented within the results, it needs to be reiterated and acknowledged that no discussion or assessment will be wholly 'objective'. In the realm of drug safety and new pharmacotherapies, there is often a trade-off between the clinical benefit and the potential risk to patients. Therapies with a favorable efficacy profile and a low risk profile, or those with a poor efficacy profile and a high-risk profile, are not contentious. What is of contention are those therapies for which the efficacy is moderate or high, and the risk to the patient is equally great. In those cases, it is important to implement well-designed trials that can clearly communicate to regulators, clinicians and patients if such a risk exists, and to assess the interplay of these risks with the potential therapeutic benefit, leading to informed decision making.

As such, the scope of the discussion will focus on the interpretation of these results, specifically considering the FDA's decision to expedite pimavanserin's approval. The following discussion will be presented first with a preface highlighting the duties of different medical experts within the realm of pharmacovigilance, followed by major safety themes amongst the results, this project's contribution to the literature, and limitations.

6.1 Duty of Medical Experts: A Preface

Amidst the discussion on trial design and regulatory policies, the most important stakeholder can sometimes be forgotten; the patient. The patient is a uniquely important stakeholder in clinical trials and their role should be highlighted. There is a dichotomy. On one hand, the patient is an expert of how they experience their condition and is in a unique position to exercise some degree of autonomy over their disease experience. They are also active agents in the development of new therapies, and clinical trials could not be conducted without their participation. On the other hand, patients are often themselves not a basic scientist, a health economist, a clinician, or a drug regulator. They are in a vulnerable position, having limited expertise required to properly exercise their autonomy. Therein lies the duty of medical experts.

The duty of a clinician is to guide the patient to make well-informed treatment decision for themselves, and to accept (or not) the therapy with the risk-benefit assessment that fits the patient's value system (Forte et al., 2018). Similarly, the duty of the regulator is as a legal authority, as an arbiter, and as a public advocate. Firstly, the regulator ensures that the information provided for the risk-benefit profile is clear enough such that the clinician can help guide the patient to make an informed treatment decision. This is done by setting the rules regarding how this efficacy and safety information is collected, presented, and analyzed. Secondly, they act as an arbiter, such that once a clear risk-benefit profile has been established, it can be analyzed and debated before being authorized to clinicians and to patients (Nelson et al., 2014). Finally, the third duty of a regulator is that of an advocate for the best interest of the public by "helping to speed

innovations that make medical products more effective, safer, and more affordable and by helping the public get the accurate, science-based information they need to use medical products and foods to maintain and improve their health" (Food and Drug Administration, 2018c).

By contrast, the duty of the drug manufacture is arguably primarily to its shareholders, which is achieved by maximizing corporate profitability of its medicines either through expanded indications or other means (Kitsis, 2011). Oftentimes, these different duties align with one another, and the patient and their family receive the appropriate care (Fleming et al., 2017). There are instances, however, when the different duties of each party are not aligned, which may result in harm to patients (Fleming et al., 2017). The role of the regulator is particularly important as their actions can impact the faith the public has towards the medical establishment.

Through this lens, the FDA as regulators in the case of pimavanserin had several responsibilities. On a systems level, the FDA needed to ensure that the drug and indication were suitable for the expedited pathways. On an approval decision level, the FDA needed to ensure there was enough quality efficacy and safety evidence presented through trials that were 'fit for purpose', such that their approval decision, if one were to be granted, was well informed. The former was not within the scope of this project, as many clinicians, patients, and their families will argue that PDP is a pressing indication with no current treatment alternatives that requires expedited treatment. What has been established through this study, however, is that there are serious methodological and risk of bias concerns.

6.2 Positive Trial Features

The most notable positive features of the RCTs were the choice of the endpoints and the perspectives from which those endpoints were elicited. The use of the SAPS-PD as the primary endpoint, along with other general psychiatric, motor, and quality of life metrics as secondary endpoints, allowed the RCTs to capture a more holistic understanding of pimavanserin's efficacy. This is especially important given that PDP is multifaceted disease, and that for pimavanserin to demonstrate superior clinical efficacy, it could not exert degenerative effects on motor functions like its dopaminergic predecessors. This was accomplished by the sponsors using the UPDRS and the MDS-UPDRS. In addition, using a broad range of secondary endpoints to capture a complex condition is especially important in late-stage trials that hold greater weight in clinical and policy decision-making, therefore requiring a more comprehensive understanding of the therapy's properties (McLeod et al., 2019).

As well, the use of patient reported outcomes (PROs) and care partner reported outcomes to capture features like ADLs and the potential impact on care partner burden is commendable. With the paradigm shift to patient-centered care and clinical trials design, incorporating metrics that report on both patients and care partners can be a means to help achieve that (Sharma, 2015). With PROs gaining prominence in the past two decades, their clinical, health economic, and regulatory benefits have also been recognized. These include: 1) helping to guide better treatment decisions; 2) enriching regulators' understanding of patients' lived experiences, especially with difficult to capture domains (e.g., pain and fatigue); 3) improving cost-effective analyses; and 4) supporting patient advocacy (Mercieca-Bebber et al., 2018). With a condition like PDP, where the impact of treatment is felt by close family members, using these metrics to communicate varying disease experiences put the FDA in a better position to understand pimavanserin's efficacy and impact on different stakeholders.

6.3 Primary Concerns

There was a wide spectrum of concerns identified in this analysis, as highlighted by the RoB 2 and PRECIS-2. These concerns can most succinctly be understood and discussed through the lens of three main themes. These are 1) inadequate number and length of trials, 2) high risk of bias, and 3) poor trial design leading to a poorly understood safety profile.

6.3.1 Inadequate Number and Length of Trials

The most prominent concern identified in this study was the inadequate number and length of the RCTs that formed the basis for regulatory approval. As previously mentioned, the FDA's own threshold established in 1998 to require at least two adequate and well-controlled studies was not met. With this benchmark, while the sponsor technically had four RCTs, and had satisfied the requirement of two studies, there were two issues that arose from this.

Firstly, only one RCT (NCT01174004) formed the basis for the FDA's approval (Food and Drug Administration, 2016). Stemming from the Breakthrough Therapy designation, the standard of evidence for pimavanserin's approval as agreed between Acadia Pharmaceuticals and the FDA rested only on one strongly positive study supported by data from other trials, rather than two studies which are each convincing on their own. This agreement can be referenced in the FDA's PDAC transcript, which is attached as *Supplementary File D* and can be referenced on page 150. Specifically, this study highlights concerns that under these expedited mechanisms, the criteria suitable for warranting approvals was not met.

Secondly, with regards to the term "adequate trials" and their impact on establishing a clinically meaningful effect, the RCTs with a trial length ranging from 6-8 weeks is arguably not long enough. The disease experience of PDP is long lasting and in in 60% of cases, leads to a patient's eventual institutionalization (Zahodne & Fernandez, 2008; Aarsland et al., 2000). As such, to paint a more comprehensive picture of pimavanserin's clinically meaningful effect, an adequate trial length extending past 6-8 weeks should have been established. While it is understandable that it is both logistically and financially unfeasible to require sponsors to implement trials that are several years in length, such trials for PDP exist. For example, the exploratory trials from the Parkinson Study Group (1999) and Pollak et al. (2004) using clozapine for the treatment of PDP had a total study period of 14 months and five months, respectively. Although these two study periods vary greatly, longer trials are clearly possible and would have improved the quality the data. it should be expected that the RCTs of therapies for the same indication be of a similar length. Even though pimavanserin was a Breakthrough Therapy designation product within the Priority Review pathway, it should not preclude the implementation of a sufficient trial period.

It is also concerning that the benefits conferred within the Breakthrough Therapy designation included intensive guidance from the FDA on the drug's development and trial design, yet each of the four RCTs were of insufficient length. This may simply suggest that the guidance from the FDA did not include a discussion with the sponsor about adopting a suitable trial length. Alternatively, it may suggest that a sufficient trial length was not a prerequisite to gaining a Breakthrough Therapy designation, or that it was not a prerequisite to gaining final regulatory approval as an expedited product. The latter, if true, would be extremely concerning, as it would mean that the threshold for final regulatory approval is different for expedited products compared to non-expedited products, and would be contrary to the FDA's prescribed messaging that the approval standards are uniform for all regulatory approval applications. At the time of this manuscript's publication, no additional systematic review has been conducted comparing the trial lengths between expedited and non-expedited products. Additional research is recommended in this area to further investigate this concern.

6.3.2 High Risk of Bias: Missing Data

A secondary concern that emerged from the results was the high risk of bias observed using the RoB 2 amongst the RCTs. One of the most notable concerns stemmed from the poorly explained nature of the missing data (i.e., randomness). This alone was enough to cause the rating for the pivotal study (NCT01174004) to receive a high risk of bias rating as opposed to a moderate risk of bias rating. Although participant attrition in clinical trials is not unexpected, different rates of losses between the experimental and control group, as observed in pimavanserin's trials, can lead to attrition bias (Bankhead et al., 2017). A review of RCTs conducted by Akl et al. (2012) estimated that up to 33% of trial outcomes were no longer significant amongst trials where the missing data were disproportionately represented in the experimental group. This is particularly concerning since the pivotal trial (NCT01174004), which formed the basis of pimavanserin's approval, did not employ any statistical compensatory techniques, and simply chose to not impute the data. With this review in mind, the systemic risk of bias originating from missing data could have negatively influenced the interpretation of pimavanserin's efficacy as a breakthrough product. As previously mentioned, an analysis of the PDAC's meeting transcript revealed that the FDA's approval of pimavanserin was primarily dependent on NCT01174004 and supported by data from earlier trials. Putting so much interpretive weight on a single trial at a high risk of bias could compromise the interpretability of pimavanserin's safety and efficacy profile.

6.3.3 Poor Trial Design and Poorly Understood Safety Profile

Another theme of concern that was uncovered by the RoB 2 and PRECIS -2 tool was the poor design of the safety studies. Although the safety studies were pragmatic in nature, as illustrated by the PRECIS-2 graphs, this was not the issue. In fact, pragmatic safety studies may be beneficial in that they allow study investigators to observe patients in a 'real-world' environment, where there may be interactions with other medications, co-morbid conditions and variable adherence regimens (Treweek & Zwarenstein, 2009). The main concern with these studies, rather, was that there was a selection bias in that participants were selected from pre-existing RCTs. As well, for NCT00550238, the comparison of mortality risk, serious adverse events, and adverse events, were made between patients on pimavanserin and concurrent atypical antipsychotics compared to patients only on pimavanserin. This made it difficult to ascertain the standalone risk of pimavanserin compared to placebo and confounds the understanding of its safety profile. Unfortunately, approving a drug with a poorly understood risk profile means that the onus

and risk of uncovering safety risks shifts from regulators to patients. This issue was echoed by members of the PDAC, who despite voting favorably on pimavanserin, brought up reservations about the unclear risk to patients.

With respect to being a Breakthrough Therapy designation product, it is concerning that a drug that was acknowledged to have a poorly understood safety profile was approved for market authorization. It raises questions as to whether the FDA is trading 'perceived benefit' for 'perceived risk', given that pimavanserin is the first therapy specifically indicated for PDP. If that is the case, then it raises even more questions as to whether the FDA ought to hold the authority to trade potential risk for potential benefit on behalf of patients. Again, this is not a point against the FDA approving drugs with a *known* harmful safety profile, as that decision would be understandable. For example, oncology drugs frequently have a toxic safety profile, yet they are still approved due to their known benefit. Rather, the concern is directed at the FDA approving drugs with an *inadequately studied* safety profile, in which the risk is not well understood, and this issue is compounded by polypharmacy within the patient demographic.

Therein also lies the issue with expedited approvals. As previously discussed, the FDA as a regulator has duties as a legal authority, an arbiter, and as a public advocate. By creating expedited pathways where drugs can be expeditiously reviewed and approved on limited safety data, potentially resulting in harm to patients, the FDA may be failing in their roles.

6.4 Contributions to the Literature

Several prominent pimavanserin safety studies have already been conducted. Some of these studies include a re-analysis conducted by the FDA using public and sponsor-provided data (Food and Drug Administration, 2018d), an independent retrospective cohort study of 2186 pimavanserin users from an administrative database (Hwang et al., 2021), and an ongoing independently conducted trial (VA Office of Research and Development, 2022). These post-hoc studies, although important, contribute narrowly to the literature by debating the nuances and statistical significance of safety signals.

This study, however, contributes to the literature by examining the very foundation of pimavanserin's approval, and raises questions regarding whether the quality of evidence, determined by the risk of bias and trial design, should have led to an FDA approval. To the knowledge of the author, this study is the first of its kind for pimavanserin.

Finally, this study contributes to the literature by highlighting the concerns of substandard trial design and risk of bias when expediting 'first-in-line' drugs. It highlights that despite stakeholder demand, there is still a duty on the part of the regulator to ensure quality evidence is being submitted and assessed. An expedited approval should not mean a compromise in quality and safety or "lowering the bar" *per se*. Rather, it should only mean that the sponsor is able to reach the threshold for generating quality efficacy and safety more quickly, with guidance and support from regulators.

94

6.5 Limitations

There are several limitations that should be considered when interpreting these results. Primarily, the scope of this study is limited to the FDA's approval of pimavanserin and to PDP, and may might not be generalizable to other disease divisions within the FDA or other drugs. With each new drug application, there may be other variables that may impact the design, feasibility of implementation, and integration of clinical trial data. As well, because pimavanserin was approved through the Breakthrough Therapy designation and Priority Review pathway, the results of this study may not be fully generalizable to the Fast-Track and Accelerated Approval designations. Future research to comprehensively assess the full scope of these expedited programs, and the robustness of their pivotal trials is recommended.

Another limitation of the study is the subjective nature of the RoB 2 and PRECIS-2 tools. No tool is perfect. As each tool requires interpretation from the individual rater using it, there is a degree of subjectivity within the reported results. This may be even more pronounced in the RoB 2 tool, as there have been concerns from users regarding its steep learning curve and poor interrater reliability (Minozzi et al., 2020). This stems from issues relating to the difficulty interpreting the signaling questions, new terminology, amongst other issues, which may hinder the application and interpretation of the tool's results (Minozzi et al., 2020). In an ideal research setting, this project would have been conducted with multiple raters with extensive expertise in clinical trials. However, practical implementation of such a project given the resources available to the research team would be untenable. A final limitation of the study is the lack of available protocols and publications for some of the studies, in particular the single-armed studies. The lack of published information made it difficult to populate certain domains of the RoB 2 and PRECIS-2 tool, and as a result, limited the interpretation of certain domains to a degree. This is also the reason why the single-arm studies were included in the results but were not contextualized in the discussion, as the author was hesitant to make premature assumptions and instead focused the narrative on the RCTs and safety studies.

6.6 Next Steps

The next steps for the medical and research community should be to continue monitoring potential safety signals, as well as to conduct independent RCTs with the aim of validating the sponsor's original safety and efficacy claims. The reason that third party investigations should be conducted is because sponsor led RCTs may be at high risk of bias, as demonstrated by this study. Although this may burden the medical and academic community, the FDA could introduce a system where fees are levied as part of the regulatory submission process and be used towards conducting confirmatory trials. These confirmatory trials would not be conducted for every submission application but may be limited to submissions where the original trials were at a high risk of bias.

The next steps for the FDA and for the broader regulatory community is threefold. The FDA should not only continue to monitor pimavanserin for safety issues, but to more stringently evaluate Acadia Pharmaceuticals' future applications to expand pimavanserin's indications, with particular attention on their pivotal clinical trials. This includes Acadia Pharmaceuticals' attempt to expand pimavanserin for the treatment of dementia-related psychosis and Alzheimer's disease psychosis.

Secondly, the FDA should re-examine the threshold for approving novel therapies through the expedited pathways, and to reconsider the potential harms posed to patients when drugs are approved on limited or biased clinical trial data. This would include standardizing the length of trials that should be expected for different indications or drugs classes. This would more easily convey the standards and expectations for study designs that the FDA have for sponsors. This would also benefit sponsors, as it means that there is less ambiguity when it comes to designing and conducting clinical trials, leading to a more transparent regulatory submission process.

Thirdly, future efforts should focus on incorporating tools to formally assess risk of bias and trial robustness into official regulatory processes. While the specific mechanisms to implement such tools are out of the scope of this paper, the RoB 2 can be highlighted as an example. Specifically, the RoB 2 could be used as a filter for drug applications. In this scenario, only drug applications with a low risk of bias would be allowed to be considered for regulatory approval, and those with some or high risk of bias would be rejected. Further research into the optimal implementation of the RoB 2 is recommended.

6.7 Conclusion

With pimavanserin, it was demonstrated that there exists a substantive risk of bias amongst the pivotal RCT and methodological design concerns with the safety studies.
Based on these findings, the conclusion to the research question is that there was not enough quality evidence and suitably designed trials to warrant an accurate safety and efficacy assessment. Specifically, an insufficient number and length of trials, poor handling of missing data, and substandard safety trials contributed to this conclusion. There is still much to be learned about the application of expedited trials, especially regarding the standards surrounding pivotal trials and the robustness of their designs. Future research should continue to monitor pimavanserin's safety, as well as provide actionable guidelines for the FDA to improve their regulatory processes moving forward to protect patients' best interests.

References

Aarsland, D., Larsen, J. P., Tandberg, E., & Laake, K. (2000). Predictors of Nursing Home Placement in Parkinson's Disease: A Population-Based, Prospective Study. *Journal of the American Geriatrics Society*. https://doi.org/10.1111/j.1532-5415.2000.tb06891.x

Acadia Pharmaceuticals. (2014, September 2). ACADIA Pharmaceuticals Receives FDA Breakthrough Therapy Designation for NUPLAZIDTM (Pimavanserin) for Parkinson's Disease Psychosis [Press release]. https://ir.acadia-pharm.com/newsreleases/news-release-details/acadia-pharmaceuticals-receives-fda-breakthroughtherapy

Acadia Pharmaceuticals. (2015, November 2). ACADIA Pharmaceuticals Announces FDA Priority Review of NUPLAZIDTM (Pimavanserin) New Drug Application for Parkinson's Disease Psychosis [Press release]. https://acadia.com/media/newsreleases/acadia-pharmaceuticals-announces-fda-priority-review-of-nuplazidpimavanserin-new-drug-application-for-parkinsons-disease-psychosis/

- Acadia Pharmaceuticals. (2016a, March 29). FDA Advisory Committee Votes 12 to 2 That Benefits of ACADIA Pharmaceuticals' NUPLAZIDTM (Pimavanserin) for the Treatment of Psychosis Associated with Parkinson's Disease Outweigh the Risks [Press release]. https://acadia.com/media/news-releases/fda-advisory-committeevotes-12-to-2-that-benefits-of-acadia-pharmaceuticals-nuplazid-pimavanserin-forthe-treatment-of-psychosis-associated-with-parkinsons-disease-outwei/
- Acadia Pharmaceuticals. (2016b, May 31). ACADIA Pharmaceuticals Announces NUPLAZIDTM (pimavanserin) Is Now Available for the Treatment of Hallucinations and Delusions Associated with Parkinson's Disease Psychosis

[Press release]. https://ir.acadia-pharm.com/news-releases/news-releasedetails/acadia-pharmaceuticals-announces-nuplazidtm-pimavanserin-now

- Acadia Pharmaceuticals. (2021, April 5). Acadia Pharmaceuticals Receives Complete Response Letter from U.S. FDA for Supplemental New Drug Application for Pimavanserin for the Treatment of Hallucinations and Delusions Associated with Dementia-Related Psychosis [Press release]. https://acadia.com/media/newsreleases/acadia-pharmaceuticals-receives-complete-response-letter-from-u-s-fdafor-supplemental-new-drug-application-for-pimavanserin-for-the-treatment-ofhallucinations-and-delusions-associated-with-dementia/
- Acadia Pharmaceuticals. (2022a, June 17). *Acadia Pharmaceuticals Announces Outcome* of FDA Advisory Committee for NUPLAZID® (pimavanserin) for the Treatment of Alzheimer's Disease Psychosis [Press release]. https://acadia.com/media/newsreleases/acadia-pharmaceuticals-announces-outcome-of-fda-advisory-committeefor-nuplazid-pimavanserin-for-the-treatment-of-alzheimers-disease-psychosis/
- Acadia Pharmaceuticals. (2022b, August 4). Acadia Pharmaceuticals Receives Complete Response Letter from U.S. FDA for Supplemental New Drug Application for Pimavanserin for the Treatment of Hallucinations and Delusions Associated with Alzheimer's Disease Psychosis [Press release]. https://acadia.com/media/newsreleases/acadia-pharmaceuticals-receives-complete-response-letter-from-u-s-fdafor-supplemental-new-drug-application-for-pimavanserin-for-the-treatment-ofhallucinations-and-delusions-associated-with-alzheime/

Acadia Pharmaceuticals. (2022c, August 9). Pipeline. https://acadia.com/pipeline/

- Adler, C. H., & Thorpy, M. J. (2005). Sleep issues in Parkinson's disease. *Neurology*, 64(Issue 12, Supplement 3), S12–S20. https://doi.org/10.1212/wnl.64.12_suppl_3.s12
- Akl, E. A., Briel, M., You, J. J., Sun, X., Johnston, B. C., Busse, J. W., Mulla, S. M., Lamontagne, F., Bassler, D., Vera, C., Alshurafa, M., Katsios, C. M., Zhou, Q., Cukierman-Yaffe, T., Gangji, A. S., Mills, E. J., Walter, S. D., Cook, D. J., Schünemann, H. J., . . . Guyatt, G. H. (2012). Potential impact on estimated treatment effects of information lost to follow-up in randomised controlled trials (LOST-IT): systematic review. *BMJ*, *344*(may18 1), e2809. https://doi.org/10.1136/bmj.e2809
- Andreason, P. (2016). *Medical Review* (Reference ID: 3882135). Center for Drug Evaluation and Research, Food and Drug Administration.
- Aronson, S., Dyke, C. M., Stierer, K. D., Levy, J. H., Cheung, A. T., Lumb, P. D.,
 Kereiakes, D. J., & Newman, M. F. (2008). The ECLIPSE Trials: Comparative
 Studies of Clevidipine to Nitroglycerin, Sodium Nitroprusside, and Nicardipine
 for Acute Hypertension Treatment in Cardiac Surgery Patients. *Anesthesia & Analgesia*, 107(4), 1110–1121. https://doi.org/10.1213/ane.0b013e31818240db
- Bankhead, C., Aronson, J., & Nunan, D. (2017). *Attrition Bias*. Catalogue of Bias Collaboration. https://catalogofbias.org/biases/attrition-bias/

Biogen. (2020, November 6). Update on FDA Advisory Committee's Meeting on Aducanumab in Alzheimer's Disease [Press release].
https://investors.biogen.com/news-releases/news-release-details/update-fdaadvisory-committees-meeting-aducanumab-alzheimers Brookings Institute. (2021). Brookings 2021 Annual Report.

https://annualreport.brookings.edu/honor-roll/

- Brown, J. D., Cicali, B., Henriksen, C., Malaty, I., Okun, M. S., & Armstrong, M. J. (2021). Comparative pharmacovigilance assessment of mortality with pimavanserin in Parkinson disease-related psychosis. *Journal of Managed Care &Amp; Specialty Pharmacy*, 27(6), 785–790.
 https://doi.org/10.18553/jmcp.2021.27.6.785
- Busner, J., & Targum, S. D. (2007). The clinical global impressions scale: applying a research tool in clinical practice. *Psychiatry (Edgmont (Pa. : Township))*.
- Canadian Agency for Drugs and Technologies in Health. (2020). CADTH Common Drug Review (Patient Input - Apomorphine Hydrochloride). In *Canadian Agency for Drugs and Technologies in Health (CADTH)*. Canadian Agency for Drugs and Technologies in Health (CADTH).

https://www.cadth.ca/sites/default/files/cdr/relatedinfo/Kynmobi%20-

%20Patient%20Group%20Input_For%20Posting.pdf

Canadian Institute for Health Information. (2007). *The Burden of Neurological Diseases, Disorders and Injuries in Canada.*

https://publications.gc.ca/collections/collection_2007/cihi-icis/H118-42-2007E.pdf

Checcucci, E., Veccia, A., De Cillis, S., Piramide, F., Volpi, G., Amparore, D., Pecoraro, A., Piana, A., Granato, S., Verri, P., Sica, M., Meziere, J., Carbonaro, B.,
Piscitello, S., Zamengo, D., Cacciamani, G., Okhunov, Z., Puliatti, S., Taratkin, M., . . . Porpiglia, F. (2021). New Ultra-minimally Invasive Surgical Treatment for Benign Prostatic Hyperplasia: A Systematic Review and Analysis of

Comparative Outcomes. European Urology Open Science.

https://doi.org/10.1016/j.euros.2021.08.009

- Chen, J. J. (2017). Treatment of psychotic symptoms in patients with Parkinson disease. *Mental Health Clinician*, 7(6), 262–270. https://doi.org/10.9740/mhc.2017.11.262
- Ciani, O., Buyse, M., Drummond, M., Rasi, G., Saad, E. D., & Taylor, R. S. (2017). Time to Review the Role of Surrogate End Points in Health Policy: State of the Art and the Way Forward. *Value in Health*, 20(3), 487–495.

https://doi.org/10.1016/j.jval.2016.10.011

CMAJ. (2005). Vioxx: lessons for Health Canada and the FDA. *Canadian Medical* Association Journal, 172(1), 5–5. https://doi.org/10.1503/cmaj.045206

Cochrane. (n.d.). About us. https://www.cochrane.org/about-us

- Cox, E. M., Edmund, A. V., Kratz, E., Lockwood, S. H., & Shankar, A. (2020).
 Regulatory Affairs 101: Introduction to Expedited Regulatory Pathways. *Clinical and Translational Science*, *13*(3), 451–461. https://doi.org/10.1111/cts.12745
- Cucherat, M., Laporte, S., Delaitre, O., Behier, J., D'Andon, A., Binlich, F., Bureau, S.,
 Cornu, C., Fouret, C., Labouret, N. H., Laviolle, B., Miadi-Fargier, H., Paoletti,
 X., Roustit, M., Danchin, N., Varoqueaux, N., Vicaut, E., & Westerloppe, J.
 (2020). From single-arm studies to externally controlled studies. Methodological
 considerations and guidelines. *Therapie*, 75(1), 21–27.
 https://doi.org/10.1016/j.therap.2019.11.007
- Dorsey, E. R., Elbaz, A., Nichols, E., Abbasi, N., Abd-Allah, F., Abdelalim, A., Adsuar,
 J. C., Ansha, M. G., Brayne, C., Choi, J. Y. J., Collado-Mateo, D., Dahodwala, N.,
 Do, H. P., Edessa, D., Endres, M., Fereshtehnejad, S. M., Foreman, K. J., Gankpe,
 F. G., Gupta, R., . . . Murray, C. J. L. (2018). Global, regional, and national

burden of Parkinson's disease, 1990–2016: a systematic analysis for the Global Burden of Disease Study 2016. *The Lancet Neurology*, *17*(11), 939–953. https://doi.org/10.1016/s1474-4422(18)30295-3

- Downing, N. S., Shah, N. D., Aminawung, J. A., Pease, A. M., Zeitoun, J. D., Krumholz, H. M., & Ross, J. S. (2017). Postmarket Safety Events Among Novel Therapeutics Approved by the US Food and Drug Administration Between 2001 and 2010. *JAMA*, *317*(18), 1854. https://doi.org/10.1001/jama.2017.5150
- Ellis, B., & Hicken, M. (2018, April 9). Reports of death spark concern about Parkinson's drug. CNN. https://edition.cnn.com/2018/04/09/health/parkinsons-drug-nuplazidinvs/index.html
- Fernandez, H. H., Aarsland, D., Fénelon, G., Friedman, J. H., Marsh, L., Tröster, A. I., Poewe, W., Rascol, O., Sampaio, C., Stebbins, G. T., & Goetz, C. G. (2008).
 Scales to assess psychosis in Parkinson's disease: Critique and recommendations. *Movement Disorders*, 23(4), 484–500. https://doi.org/10.1002/mds.21875
- Fernandez, H. H., Donnelly, E. M., & Friedman, J. H. (2004). Long-term outcome of clozapine use for psychosis in parkinsonian patients. *Movement Disorders*, 19(7), 831–833. https://doi.org/10.1002/mds.20051
- Feuerstein, A., Herper, M., & Garde, D. (2022, January 12). Inside 'Project Onyx': How Biogen used an FDA back channel to win approval of its polarizing Alzheimer's drug. STAT. https://www.statnews.com/2021/06/29/biogen-fda-alzheimers-drugapproval-aduhelm-project-onyx/
- Fleming, T. R., & DeMets, D. L. (1996). Surrogate End Points in Clinical Trials: Are We Being Misled? Annals of Internal Medicine, 125(7), 605. https://doi.org/10.7326/0003-4819-125-7-199610010-00011

- Fleming, T. R., DeMets, D. L., & McShane, L. M. (2017). Discussion: The role, position, and function of the FDA—The past, present, and future. *Biostatistics*. https://doi.org/10.1093/biostatistics/kxx023
- Food and Drug Administration. (1998). Providing Clinical Evidence of Effectiveness for Human Drug and Biological Products. In *fda.gov*. Food and Drug Administration. https://www.fda.gov/regulatory-information/search-fda-guidancedocuments/providing-clinical-evidence-effectiveness-human-drug-and-biologicalproducts
- Food and Drug Administration. (2001). E10 Choice of Control Group and Related Issues in Clinical Trials. In *fda.gov* (FDA-1999-D-1874).

https://www.fda.gov/regulatory-information/search-fda-guidance-documents/e10choice-control-group-and-related-issues-clinical-trials

Food and Drug Administration. (2016, March 29). Transcript for the March 29, 2016 Meeting of the Psychopharmacologic Drugs Advisory Committee (PDAC). http://wayback.archive-

it.org/7993/20170113060919/http://www.fda.gov/downloads/AdvisoryCommittee s/CommitteesMeetingMaterials/Drugs/PsychopharmacologicDrugsAdvisoryCom mittee/UCM502050.pdf

- Food and Drug Administration. (2017, November 14). Drugs@FDA Glossary of Terms. U.S. Food And Drug Administration. https://www.fda.gov/drugs/drug-approvalsand-databases/drugsfda-glossary-terms
- Food and Drug Administration. (2018a, January 4). *Step 4: FDA Drug Review*. U.S. Food And Drug Administration. https://www.fda.gov/patients/drug-developmentprocess/step-4-fda-drug-review

 Food and Drug Administration. (2018b, February 23). Fast Track, Breakthrough Therapy, Accelerated Approval, Priority Review. U.S. Food And Drug Administration. https://www.fda.gov/patients/learn-about-drug-and-deviceapprovals/fast-track-breakthrough-therapy-accelerated-approval-priority-review
 Food and Drug Administration. (2018c, March 28). What We Do.

https://www.fda.gov/about-fda/what-we-

do#:~:text=The%20Food%20and%20Drug%20Administration,and%20products%
20that%20emit%20radiation.

Food and Drug Administration. (2018d, September 20). FDA analysis finds no new or unexpected safety risks associated with Nuplazid (pimavanserin), a medication to treat the hallucinations and delusions of Parkinson's disease psychosis [Press release]. https://www.fda.gov/drugs/drug-safety-and-availability/fda-analysisfinds-no-new-or-unexpected-safety-risks-associated-nuplazid-pimavanserinmedication

Food and Drug Administration. (2019, April 23). *Promoting Safe & Effective Drugs for 100 Years*. U.S. Food And Drug Administration. https://www.fda.gov/aboutfda/histories-product-regulation/promoting-safe-effective-drugs-100-years

Food and Drug Administration. (2020, February 24). *Therapeutic Biologics Applications* (*BLA*). U.S. Food And Drug Administration. https://www.fda.gov/drugs/types-applications/therapeutic-biologics-applications-bla

Food and Drug Administration. (2021a, June 7). *FDA Grants Accelerated Approval for Alzheimer's Drug* [Press release]. https://www.fda.gov/news-events/pressannouncements/fda-grants-accelerated-approval-alzheimers-drug Food and Drug Administration. (2021b, October 23). FDA Adverse Event Reporting System (FAERS) Public Dashboard. fda.gov.

https://www.fda.gov/drugs/questions-and-answers-fdas-adverse-event-reporting-drugs/questions-and-answers-fdas-adverse-event-reporting-drugs/questions-and-answers-fdas-adverse-event-reporting-drugs/questions-and-answers-fdas-adverse-event-reporting-drugs/questions-and-answers-fdas-adverse-event-reporting-drugs/questions-and-answers-fdas-adverse-event-reporting-drugs/questions-and-answers-fdas-adverse-event-reporting-drugs/questions-and-answers-fdas-adverse-event-reporting-drugs/questions-adverse-event-reporting-drugs/questions-adverse-event-reporting-drugs/questions-adverse-event-reporting-drugs/questions-adverse-event-reporting-drugs/questions-adverse-event-reporting-drugs/questions-adverse-event-reporting-drugs/questions-adverse-event-reporting-drugs/questions-adverse-event-reporting-drugs/questions-adverse-event-reporting-drugs/questions-adverse-event-reporting-drugs/questions-adverse-event-reporting-drugs/questions-adverse-event-reporting-drugs/questions-drugs/questions-drugs/questions-drugs/questions-drugs/questions-drugs/questions-event-reporting-drugs/questions-drugs-

system-faers/fda-adverse-event-reporting-system-faers-public-dashboard

- Food and Drug Administration. (2022a, January 14). *Abbreviated New Drug Application* (*ANDA*). U.S. Food And Drug Administration. https://www.fda.gov/drugs/typesapplications/abbreviated-new-drug-application-anda
- Food and Drug Administration. (2022b, January 21). *New Drug Application (NDA)*. U.S. Food And Drug Administration. https://www.fda.gov/drugs/typesapplications/new-drug-application-nda
- Food and Drug Administration. (2022c, February 18). *New Drug Development and Review Process*. U.S. Food And Drug Administration.

https://www.fda.gov/drugs/cder-small-business-industry-assistance-sbia/newdrug-development-and-review-process

- Food and Drug Administration. (2022d, July 20). Investigational New Drug (IND) Application. U.S. Food And Drug Administration. https://www.fda.gov/drugs/types-applications/investigational-new-drug-indapplication
- Forte, D. N., Kawai, F., & Cohen, C. (2018). A bioethical framework to guide the decision-making process in the care of seriously ill patients. *BMC Medical Ethics*, 19(1). https://doi.org/10.1186/s12910-018-0317-y
- Frakt, A. B. (2018). The Risks and Benefits of Expedited Drug Reviews. JAMA, 320(3), 225. https://doi.org/10.1001/jama.2018.8262

Friedman, J. H. (2010). Parkinson's disease psychosis 2010: A review article. Parkinsonism &Amp; Related Disorders, 16(9), 553–560. https://doi.org/10.1016/j.parkreldis.2010.05.004

Friends of Cancer Research. (2020). *IRS Form 990*. https://friendsofcancerresearch.org/financials-policies/

Goetz, C. G., Tilley, B. C., Shaftman, S. R., Stebbins, G. T., Fahn, S., Martinez-Martin, P., Poewe, W., Sampaio, C., Stern, M. B., Dodel, R., Dubois, B., Holloway, R. G., Jankovic, J., Kulisevsky, J., Lang, A. E., Lees, A. J., Leurgans, S., LeWitt, P. A., Nyenhuis, D. L., . . . Zweig, R. A. (2008). Movement Disorder Society-sponsored revision of the Unified Parkinson's Disease Rating Scale (MDS-UPDRS): Scale presentation and clinimetric testing results. *Movement Disorders*, *23*(15), 2129–2170. https://doi.org/10.1002/mds.22340

Graham, D. J., Campen, D. H., Hui, R. L., Spence, M. M., Cheetham, C., Levy, G. N., Shoor, S., & Ray, W. A. (2005). Risk of acute myocardial infarction and sudden cardiac death in patients treated with cyclo-oxygenase 2 selective and nonselective non-steroidal anti-inflammatory drugs: nested case-control study. *The Lancet*, 365(9458), 475–481. https://doi.org/10.1016/s0140-6736(05)17864-7

Grimes, D., Fitzpatrick, M., Gordon, J., Miyasaki, J., Fon, E. A., Schlossmacher, M.,
Suchowersky, O., Rajput, A., Lafontaine, A. L., Mestre, T., Appel-Cresswell, S.,
Kalia, S. K., Schoffer, K., Zurowski, M., Postuma, R. B., Udow, S., Fox, S.,
Barbeau, P., & Hutton, B. (2019). Canadian guideline for Parkinson disease. *Canadian Medical Association Journal*, *191*(36), E989–E1004.
https://doi.org/10.1503/cmaj.181504

- Gupta, M. A., Simpson, F., & Lyons, D. C. A. (2016). The effect of treating obstructive sleep apnea with positive airway pressure on depression and other subjective symptoms: A systematic review and meta-analysis. *Sleep Medicine Reviews*. https://doi.org/10.1016/j.smrv.2015.07.002
- Gyawali, B., Hey, S. P., & Kesselheim, A. S. (2019). Assessment of the Clinical Benefit of Cancer Drugs Receiving Accelerated Approval. JAMA Internal Medicine, 179(7), 906. https://doi.org/10.1001/jamainternmed.2019.0462
- Haddad, H. W., Malone, G. W., Comardelle, N. J., Degueure, A. E., Poliwoda, S., Kaye,
 R. J., Murnane, K. S., Kaye, A. M., & Kaye, A. D. (2022). Aduhelm, a novel antiamyloid monoclonal antibody, for the treatment of Alzheimer's Disease: A comprehensive review. *Health Psychology Research*, *10*(2). https://doi.org/10.52965/001c.37023
- Hermanowicz, N., & Edwards, K. (2015). Parkinson's disease psychosis: symptoms, management, and economic burden. *The American Journal of Managed Care*, 21(10 Suppl), s199-206.
- Hirsch, L., Jette, N., Frolkis, A., Steeves, T., & Pringsheim, T. (2016). The Incidence of Parkinson's Disease: A Systematic Review and Meta-Analysis. *Neuroepidemiology*, 46(4), 292–300. https://doi.org/10.1159/000445751

Hwang, Y. J., Alexander, G., An, H., Moore, T. A., & Mehta, H. B. (2021). Risk of Hospitalization and Death Associated With Pimavanserin Use in Older Adults With Parkinson Disease. *Neurology*, 97(13), e1266–e1275. https://doi.org/10.1212/wnl.000000000012601 Institute for Safe Medication Practices. (2017). Safety Signals for Two Novel Drugs. In Institute for Safe Medication Practices.

https://www.ismp.org/sites/default/files/attachments/2018-01/2017Q1_0.pdf

- Institute For Safe Medication Practices. (2020, September 1). *Institute For Safe Medication Practices (About Us)*. Institute for Safe Medication Practices. https://www.ismp.org/about
- Jette, A. M., Davies, A. R., Cleary, P. D., Calkins, D. J., Rubenstein, L. V., Fink, A., Kosecoff, J., Young, R. T., Brook, R. H., & Delbanco, T. L. (1986). The functional status questionnaire. *Journal of General Internal Medicine*, 1(3), 143– 149. https://doi.org/10.1007/bf02602324
- Johnson, J. R., Ning, Y. M., Farrell, A., Justice, R., Keegan, P., & Pazdur, R. (2011). Accelerated Approval of Oncology Products: The Food and Drug Administration Experience. *JNCI: Journal of the National Cancer Institute*, *103*(8), 636–644. https://doi.org/10.1093/jnci/djr062
- Kalilani, L., Asgharnejad, M., Palokangas, T., & Durgin, T. (2016). Comparing the Incidence of Falls/Fractures in Parkinson's Disease Patients in the US Population. *PLOS ONE*, *11*(9), e0161689. https://doi.org/10.1371/journal.pone.0161689
- Kemp, R., & Prasad, V. (2017). Surrogate endpoints in oncology: when are they acceptable for regulatory and clinical decisions, and are they currently overused? *BMC Medicine*, 15(1). https://doi.org/10.1186/s12916-017-0902-9
- Kepplinger, E. E. (2015). FDA's Expedited Approval Mechanisms for New Drug Products. *Biotechnology Law Report*, 34(1), 15–37. https://doi.org/10.1089/blr.2015.9999

- Kim, C., & Prasad, V. (2015). Cancer Drugs Approved on the Basis of a Surrogate End Point and Subsequent Overall Survival. JAMA Internal Medicine, 175(12), 1992. https://doi.org/10.1001/jamainternmed.2015.5868
- Kitsis, E. A. (2011). The Pharmaceutical Industry's Role in Defining Illness. *AMA Journal of Ethics*, *13*(12), 906–911.

https://doi.org/10.1001/virtualmentor.2011.13.12.oped1-1112

- Knopman, D. S., Jones, D. T., & Greicius, M. D. (2020). Failure to demonstrate efficacy of aducanumab: An analysis of the EMERGE and ENGAGE trials as reported by Biogen, December 2019. *Alzheimer's &Amp; Dementia*, *17*(4), 696–701. https://doi.org/10.1002/alz.12213
- Lang, A. E., & Lozano, A. M. (1998). Parkinson's Disease. *New England Journal of Medicine*, *339*(15), 1044–1053. https://doi.org/10.1056/nejm199810083391506
- Lenzer, J., & Brownlee, S. (2021). Should regulatory authorities approve drugs based on surrogate endpoints? *BMJ*, n2059. https://doi.org/10.1136/bmj.n2059
- Lexchin, J. (2011). Those Who Have the Gold Make the Evidence: How the Pharmaceutical Industry Biases the Outcomes of Clinical Trials of Medications. *Science and Engineering Ethics*, 18(2), 247–261. https://doi.org/10.1007/s11948-011-9265-3

 Lexchin, J., Graham, J. E., Herder, M., Jefferson, T., & Lemmens, T. (2021). Regulators, Pivotal Clinical Trials, and Drug Regulation in the Age of COVID-19.
 International Journal of Health Services. https://doi.org/10.1177/0020731420979824

- Liang, B. A. (2002). FDA use of the black box warning: time for reevaluation as a safety tool. *Journal of Clinical Anesthesia*, 14(8), 561–563. https://doi.org/10.1016/s0952-8180(02)00444-0
- Little, R. J., D'Agostino, R., Cohen, M. L., Dickersin, K., Emerson, S. S., Farrar, J. T.,
 Frangakis, C., Hogan, J. W., Molenberghs, G., Murphy, S. A., Neaton, J. D.,
 Rotnitzky, A., Scharfstein, D., Shih, W. J., Siegel, J. P., & Stern, H. (2012). The
 Prevention and Treatment of Missing Data in Clinical Trials. *New England Journal of Medicine*, *367*(14), 1355–1360. https://doi.org/10.1056/nejmsr1203730
- Loudon, K., Treweek, S., Sullivan, F., Donnan, P. T., Thorpe, K. E., & Zwarenstein, M. (2015). The PRECIS-2 tool: designing trials that are fit for purpose. *BMJ*, *350*(may08 1), h2147. https://doi.org/10.1136/bmj.h2147
- Loudon, K., Zwarenstein, M., Sullivan, F., Donnan, P. T., Gágyor, I., Hobbelen, H., Althabe, F., Krishnan, J. A., & Treweek, S. (2017). The PRECIS-2 tool has good interrater reliability and modest discriminant validity. *Journal of Clinical Epidemiology*, 88, 113–121. https://doi.org/10.1016/j.jclinepi.2017.06.001
- Mahase, E. (2021). FDA allows drugs without proven clinical benefit to languish for years on accelerated pathway. *BMJ*, n1898. https://doi.org/10.1136/bmj.n1898
- Marinus, J., Visser, M., Van Hilten, J. J., Lammers, G. J., & Stiggelbout, A. M. (2003).
 Assessment of Sleep and Sleepiness in Parkinson Disease. *Sleep*, 26(8), 1049–1054. https://doi.org/10.1093/sleep/26.8.1049
- Mathis, M. V., Muoio, B. M., Andreason, P., Avila, A. M., Farchione, T., Atrakchi, A., & Temple, R. J. (2017). The US Food and Drug Administration's Perspective on the New Antipsychotic Pimavanserin. *The Journal of Clinical Psychiatry*, 78(06), e668–e673. https://doi.org/10.4088/jcp.16r11119

- McLean, G., Hindle, J. V., Guthrie, B., & Mercer, S. W. (2017). Co-morbidity and polypharmacy in Parkinson's disease: insights from a large Scottish primary care database. *BMC Neurology*, 17(1). https://doi.org/10.1186/s12883-017-0904-4
- McLeod, C., Norman, R., Litton, E., Saville, B. R., Webb, S., & Snelling, T. L. (2019).
 Choosing primary endpoints for clinical trials of health care interventions. *Contemporary Clinical Trials Communications*, 16, 100486.
 https://doi.org/10.1016/j.conctc.2019.100486
- Mercieca-Bebber, R., King, M., Calvert, M., Stockler, M. R., & Friedlander, M. (2018). The importance of patient-reported outcomes in clinical trials and strategies for future optimization. *Patient Related Outcome Measures*, *Volume 9*, 353–367. https://doi.org/10.2147/prom.s156279
- Minozzi, S., Cinquini, M., Gianola, S., González-Lorenzo, M., & Banzi, R. (2020). The revised Cochrane risk of bias tool for randomized trials (RoB 2) showed low interrater reliability and challenges in its application. *Journal of Clinical Epidemiology*, *126*, 37–44. https://doi.org/10.1016/j.jclinepi.2020.06.015
- Minozzi, S., Dwan, K., Borrelli, F., & Filippini, G. (2022). Reliability of the revised
 Cochrane risk-of-bias tool for randomised trials (RoB2) improved with the use of
 implementation instruction. *Journal of Clinical Epidemiology*, *141*, 99–105.
 https://doi.org/10.1016/j.jclinepi.2021.09.021
- Mlinarić, A., Horvat, M. T., & Smolčić, V. (2017). Dealing with the positive publication bias: Why you should really publish your negative results. *Biochemia Medica*, 27(3). https://doi.org/10.11613/bm.2017.030201
- Moher, D., Jadad, A. R., Nichol, G., Penman, M., Tugwell, P., & Walsh, S. (1995). Assessing the quality of randomized controlled trials: An annotated bibliography

of scales and checklists. *Controlled Clinical Trials*, *16*(1), 62–73. https://doi.org/10.1016/0197-2456(94)00031-w

- Moskovitz, C., Moses, H., 3rd, & Klawans, H. L. (1978). Levodopa-induced psychosis: a kindling phenomenon. *American Journal of Psychiatry*, 135(6), 669–675. https://doi.org/10.1176/ajp.135.6.669
- National Library of Medicine (NLM). (2020, April 6). National Institutes of Health (NIH). https://www.nih.gov/about-nih/what-we-do/nih-almanac/national-librarymedicine-nlm
- Nelson, L. S., Loh, M. S., & Perrone, J. (2014). Assuring Safety of Inherently Unsafe Medications: the FDA Risk Evaluation and Mitigation Strategies. *Journal of Medical Toxicology*. https://doi.org/10.1007/s13181-013-0374-z

Neurological Health Charities Canada, Public Health Agency of Canada, Health Canada, & The Canadian Institutes of Health Research. (2014). *Mapping Connections Understanding of Neurological Conditions in Canada*. Public Health Agency of Canada. https://www.canada.ca/content/dam/phac-aspc/migration/phacaspc/publicat/cd-mc/mc-ec/assets/pdf/mc-ec-eng.pdf

Nguyen, T. P. P., Thibault, D., Hamedani, A. G., Weintraub, D., & Willis, A. W. (2022). Atypical antipsychotic use and mortality risk in Parkinson disease. *Parkinsonism* &*Amp; Related Disorders*, *103*, 17–22.

https://doi.org/10.1016/j.parkreldis.2022.08.013

Page, M. J., McKenzie, J. E., Bossuyt, P. M., Boutron, I., Hoffmann, T., Mulrow, C. D.,
Shamseer, L., Tetzlaff, J., Akl, E. A., Brennan, S. E., Chou, R., Glanville, J.,
Grimshaw, J. M., Hróbjartsson, A., Lalu, M. M., Li, T., Loder, E., Mayo-Wilson,
E., McDonald, S., . . . Moher, D. (2021). The PRISMA 2020 statement: an

updated guideline for reporting systematic reviews. BMJ, n71.

https://doi.org/10.1136/bmj.n71

- Parkinson Study Group. (1999). Low-Dose Clozapine for the Treatment of Drug-Induced
 Psychosis in Parkinson's Disease. *New England Journal of Medicine*, 340(10),
 757–763. https://doi.org/10.1056/nejm199903113401003
- Patented Medicine Prices Review Board (PMPRB). (2018). MEDS Entry Watch 2016. In Patented Medicine Prices Review Board. https://publications.gc.ca/collections/collection_2018/cepmb-pmprb/H79-12-2016-eng.pdf
- Pham Nguyen, T. P., Abraham, D. S., Thibault, D., Weintraub, D., & Willis, A. W. (2021). Low continuation of antipsychotic therapy in Parkinson disease – intolerance, ineffectiveness, or inertia? *BMC Neurology*, 21(1). https://doi.org/10.1186/s12883-021-02265-x
- Pollak, P., Tison, F., Rascol, O., Destée, A., Péré, J. J., Senard, J. M., Durif, F., &
 Bourdeix, I. (2004). Clozapine in drug induced psychosis in Parkinson's disease: a randomised, placebo controlled study with open follow up. *Journal of Neurology, Neurosurgery &Amp; Psychiatry*, 75(5), 689–695.
 https://doi.org/10.1136/jnnp.2003.029868
- Redberg, R. F. (2015). Faster Drug Approvals Are Not Always Better and Can Be Worse. JAMA Internal Medicine, 175(8), 1398.

https://doi.org/10.1001/jamainternmed.2015.2857

Reich, S. G., & Savitt, J. M. (2019). Parkinson's Disease. *Medical Clinics of North America*, 103(2), 337–350. https://doi.org/10.1016/j.mcna.2018.10.014

- Sahli, Z. T., & Tarazi, F. I. (2017). Pimavanserin: novel pharmacotherapy for Parkinson's disease psychosis. *Expert Opinion on Drug Discovery*, 13(1), 103–110. https://doi.org/10.1080/17460441.2018.1394838
- Schrag, A., Hovris, A., Morley, D., Quinn, N., & Jahanshahi, M. (2006). Caregiverburden in parkinson's disease is closely associated with psychiatric symptoms, falls, and disability. *Parkinsonism &Amp; Related Disorders*, 12(1), 35–41. https://doi.org/10.1016/j.parkreldis.2005.06.011
- Sharfstein, J. M., Miller, J. H., Davis, A. C., Ross, J. S., McCarthy, M. M., Smith, B. J.,
 Chaudhry, A., Alexander, G., & Kesselheim, A. S. (2017). Blueprint for
 Transparency at the U.S. Food and Drug Administration: Recommendations to
 Advance the Development of Safe and Effective Medical Products. *Journal of Law Medicine & Ethics*, 45(S2), 7–23.

https://doi.org/10.1177/1073110517750615

- Sharma, N. (2015). Patient centric approach for clinical trials: Current trend and new opportunities. *Perspectives in Clinical Research*, 6(3), 134. https://doi.org/10.4103/2229-3485.159936
- Singh, S., & Loke, Y. K. (2012). Drug safety assessment in clinical trials: methodological challenges and opportunities. *Trials*, 13(1). https://doi.org/10.1186/1745-6215-13-138
- Soogrim, V., Ruberto, V. L., Murrough, J., & Jha, M. K. (2021). Spotlight on
 Pimavanserin Tartrate and Its Therapeutic Potential in the Treatment of Major
 Depressive Disorder: The Evidence to Date. *Drug Design, Development and Therapy, Volume 15*, 151–157. https://doi.org/10.2147/dddt.s240862

- Soumerai, S. B., Avorn, J., Gortmaker, S., & Hawley, S. (1987). Effect of government and commercial warnings on reducing prescription misuse: the case of propoxyphene. *American Journal of Public Health*, 77(12), 1518–1523. https://doi.org/10.2105/ajph.77.12.1518
- Sterne, J. a. C., Savović, J., Page, M. J., Elbers, R. G., Blencowe, N. S., Boutron, I.,
 Cates, C. J., Cheng, H., Corbett, M. A., Eldridge, S., Emberson, J., Hernán, M. A.,
 Hopewell, S., Hróbjartsson, A., Junqueira, D. R., Jüni, P., Kirkham, J. J.,
 Lasserson, T. J., Li, T., . . . Higgins, J. P. T. (2019). RoB 2: a revised tool for
 assessing risk of bias in randomised trials. *BMJ*, 14898.
 https://doi.org/10.1136/bmj.14898
- Tampi, R. R., Forester, B. P., & Agronin, M. (2021). Aducanumab: evidence from clinical trial data and controversies. *Drugs in Context*, 10, 1–9. https://doi.org/10.7573/dic.2021-7-3

Tariot, P. N., Cummings, J. L., Soto-Martin, M. E., Ballard, C., Erten-Lyons, D., Sultzer, D. L., Devanand, D. P., Weintraub, D., McEvoy, B., Youakim, J. M., Stankovic, S., & Foff, E. P. (2021). Trial of Pimavanserin in Dementia-Related Psychosis. *New England Journal of Medicine*, *385*(4), 309–319. https://doi.org/10.1056/nejmoa2034634

- Thomas, A. A., & Friedman, J. H. (2010). Current Use of Clozapine in Parkinson Disease and Related Disorders. *Clinical Neuropharmacology*, 33(1), 14–16. https://doi.org/10.1097/wnf.0b013e3181c47168
- Treweek, S., & Zwarenstein, M. (2009). Making trials matter: pragmatic and explanatory trials and the problem of applicability. *Trials*, 10(1). https://doi.org/10.1186/1745-6215-10-37

- U.S. Department of Health and Human Service, Food and Drug Administration, Center for Drug Evaluation and Research (CDER), Center for Biologics Evaluation and Research (CBER), & Oncology Center of Excellence (OCE). (2022).
 Considerations for Rescinding Breakthrough Therapy Designation Guidance for Industry. https://www.fda.gov/media/159359/download
- U.S. Department of Justice. (2006). Internal Memorandum (Proposed Indictment of Purdue Pharma LP, The Purdue Frederick Company, MIchael Friedman (COO), Howard R. Udell (EVP GC), Paul D. Goldenheim (EVP)).
- US. Government Accountability Office. (2015). FDA Expedites Many Applications, But Data for Postapproval Oversight Need Improvement. https://www.gao.gov/products/gao-16-192
- U.S. National Library of Medicine. (2005, December 7). Treatment of Hallucinosis/Psychosis in Parkinson's Disease by an Investigational Drug - Full Text View - ClinicalTrials.gov. clinicaltrials.gov.
 https://clinicaltrials.gov/ct2/show/NCT00087542
- U.S. National Library of Medicine. (2014, March 26). A Study of the Safety and Efficacy of Pimavanserin in Patients With Parkinson's Disease Psychosis Full Text View ClinicalTrials.gov. clinicaltrials.gov.

https://clinicaltrials.gov/ct2/show/NCT01174004

 U.S. National Library of Medicine. (2017a, May 17). A Study of the Safety and Efficacy of Pimavanserin (ACP-103) in Patients With Parkinson's Disease Psychosis - Full Text View - ClinicalTrials.gov. clinicaltrials.gov. https://clinicaltrials.gov/ct2/show/NCT00477672

- U.S. National Library of Medicine. (2017b, May 19). A Study of Safety and Efficacy of Pimavanserin (ACP-103) in Patients With Parkinson's Disease Psychosis - Full Text View - ClinicalTrials.gov. clinicaltrials.gov. https://clinicaltrials.gov/ct2/show/NCT00658567
- U.S. National Library of Medicine. (2019, June 24). A Study of the Safety and Tolerability of Pimavanserin (ACP-103) in Patients With Parkinson's Disease Psychosis -Study Results - ClinicalTrials.gov. clinicaltrials.gov.
 https://clinicaltrials.gov/ct2/show/results/NCT00550238
- U.S. National Library of Medicine. (2020a, August 31). Safety and Efficacy of Pimavanserin in Adults With Parkinson's Disease and Depression - Full Text View - ClinicalTrials.gov. clinicaltrials.gov. https://clinicaltrials.gov/ct2/show/NCT03482882
- U.S. National Library of Medicine. (2020b, November 23). An Open-label Safety Study of Pimavanserin in Parkinson's Disease Patients - Full Text View -ClinicalTrials.gov. clinicaltrials.gov.

https://clinicaltrials.gov/ct2/show/study/NCT01518309

- U.S. National Library of Medicine. (2022, September 1). Open-Label Study With Pimavanserin on Activities of Daily Living in Subjects With Parkinson's Disease Psychosis - Full Text View - ClinicalTrials.gov. clinicaltrials.gov. https://clinicaltrials.gov/ct2/show/NCT04292223
- VA Office of Research and Development. (2022). Pimavanserin vs. Quetiapine for Treatment of Parkinson's Psychosis (C-SAPP) [Dataset]. Clincaltrials.gov. https://clinicaltrials.gov/ct2/show/NCT04373317

- Vanover, K. E., Van Kammen, D. P., Tole, B., Schlienger, N., Williams, H., Davis, R. E.,
 Weiner, D. M., & Mayall, T. E. (2008). *International Application Published Under the Patent Cooperation Treaty* (Patent No. WO 2008/144665 A1). World
 Intellectual Property Organization.
- Voss, T., Bahr, D., Cummings, J. L., Mills, R. M., Ravina, B., & Williams, H. (2013).
 Performance of a shortened Scale for Assessment of Positive Symptoms for
 Parkinson's disease psychosis. *Parkinsonism & Related Disorders*, *19*(3), 295–299. https://doi.org/10.1016/j.parkreldis.2012.10.022
- Wagner, A. K., Chan, K. A., Dashevsky, I., Raebel, M. A., Andrade, S. E., Lafata, J. E., Davis, R. L., Gurwitz, J. H., Soumerai, S. B., & Platt, R. (2006). FDA drug prescribing warnings: is the black box half empty or half full? *Pharmacoepidemiology and Drug Safety*, *15*(6), 369–386.
 https://doi.org/10.1002/pds.1193
- Whitstock, M. T. (2017). Manufacturing the truth: from designing clinical trials to publishing trial data. *Indian Journal of Medical Ethics*. https://doi.org/10.20529/ijme.2017.096
- Williams, A. (1990). EuroQol a new facility for the measurement of health-related quality of life. *Health Policy*, *16*(3), 199–208. https://doi.org/10.1016/0168-8510(90)90421-9
- Zahodne, L. B., & Fernandez, H. H. (2008). Course, Prognosis, and Management of Psychosis in Parkinson's Disease: Are Current Treatments Really Effective? CNS Spectrums, 13(S4), 26–33. https://doi.org/10.1017/s1092852900017296

Zarit, S. H., Reever, K. E., & Bach-Peterson, J. (1980). Relatives of the Impaired Elderly:
Correlates of Feelings of Burden. *Gerontologist*, 20(6), 649–655.
https://doi.org/10.1093/geront/20.6.649

Curriculum Vitae

Name:	Ka Long Benson Law
Post-secondary Education and Degrees	University of Western Ontario London, Ontario, Canada 2021-2023, MSc
	University of Western Ontario London, Ontario, Canada 2017-2021, BHSc
Honours and Awards:	Parkinson's Society of Southwestern Ontario Graduate Student Scholarship 2022-2023
	The Canadian Institutes of Health Research (CIHR) Master's Scholarship (CGSM) 2021-2022
	Ontario Graduate Scholarship (OGS) 2021-2022, 2022-2023
Related Work Experience	Teaching Assistant (Health Policy, Health Economics, Critical Appraisal) The University of Western Ontario 2021-2023
	Health Economics and Health Technology Assessment Intern Johnson & Johnson (Janssen Pharmaceutical Company) 2022-2023

Publications:

No relevant publications to declare.